

ERD TECHNICAL NOTE No. 20

Tourism for Pro-Poor and Sustainable Growth: Economic Analysis of Tourism Projects

TUN LIN AND FRANKLIN D. DE GUZMAN

JANUARY 2007

Tun Lin is an economist and Franklin D. De Guzman is an economics officer in the Economic Analysis and Operations Support Division of the Economics and Research Department, Asian Development Bank (ADB). The authors thank colleagues at ADB and reviewers at the School of Hotel and Tourism Management, The Hong Kong Polytechnic University, for providing helpful comments on earlier drafts: Juzhong Zhuang, Yong Ye, Alfredo Perdiguero, and Wantanee Suntikul. Adora Navarro provided valuable research assistance.

Asian Development Bank

Asian Development Bank 6 ADB Avenue, Mandaluyong City 1550 Metro Manila, Philippines www.adb.org/economics

©2007 by Asian Development Bank January 2007 ISSN 1655-5236

The views expressed in this paper are those of the author(s) and do not necessarily reflect the views or policies of the Asian Development Bank.

FOREWORD

The ERD Technical Note Series deals with conceptual, analytical, or methodological issues relating to project/program economic analysis or statistical analysis. Papers in the Series are meant to enhance analytical rigor and quality in project/program preparation and economic evaluation, and improve statistical data and development indicators. ERD Technical Notes are prepared mainly, but not exclusively, by staff of the Economics and Research Department, their consultants, or resource persons primarily for internal use, but may be made available to interested external parties.

CONTENTS

Abstr	act		vii				
I.	INTR	ODUCTION	1				
	A. B.	The Rise of Pro-Poor Tourism ADB Tourism Projects	1 4				
11.		NALYTICAL FRAMEWORK AND THEORETICAL PRACTICAL CHALLENGES	5				
	A. B. C.	An Analytical Framework for Economic Analysis of Tourism Projects Theoretical Challenges Practical Challenges	5 7 8				
III.	INTE	ESTABLISHING PROJECT RATIONALE FOR PUBLIC SECTOR INTERVENTION—MARKET FAILURE ARGUMENTS AND PRO-POOR PRINCIPLE					
	A. B. C.	Building Infrastructure—Coordination Failure and Public Goods Preserving Natural and Cultural Heritage Demonstrating Good Practices and Sustainable Models— Information Externality	9				
IV.	DEN	IAND ANALYSIS: TOURIST PROFILE AND EXPENDITURES	10				
٧.	EST	IMATING TOURISM BENEFITS: TWO ALTERNATIVES	13				
	A. B. C. D.	Estimating Benefits: Economic Impact Analysis versus Cost-Benefit Analysis Standard Multiplier Analysis Econometric Techniques Input-Output Analysis	13 15 16 17				
	E.	SAM Analysis and CGE Model	18				

	A. Distributional Analysis	
	B. Financial and Institutional Sustainability	
	C. Environmental Sustainability and Preservation Principle	
VII.	AN ILLUSTRATIVE EXAMPLE: SAMBOR PREY KUK	
	CULTURAL TOURISM DEVELOPMENT PROJECT	
	A. Site Description	
	B. Project Description	
	C. Project Rationale	
	D. Demand Analysis: Visitor Profile and Expenditure	
	E. Application of Income Multiplier	
	F. Distribution Analysis	
	G. Sustainability Analysis	
VIII.	CONCLUSIONS	
RFFF	RENCES	

ABSTRACT

Despite the increasing importance of tourism in economic development and the rise of "pro-poor" tourism development strategies, properly designing and implementing tourism projects remain generally a difficult process. There are both theoretical and practical challenges in justifying public sector investments in tourism and properly measuring the projects' benefits and sustainability. There is a need to come up with an analytical framework that would address these challenges and help evaluate a tourism project's economic viability. This technical note introduces a simple framework that could underpin the systematic economic analysis of tourism projects.

I. INTRODUCTION

Tourism is one of the world's fastest growing industries (Mowforth and Munt 1998, Goodwin 2000), and an integral part of economic development strategies in developing nations since the 1960s. According to the World Tourism Organization (WTO), tourism contributes over 2% of gross domestic product (GDP) and 5% of exports to the economies of 11 of the 12 developing countries that are home to 80% of the world's poor (WTO 1998 in Ashley et al. 2000). The industry's potential to generate foreign exchange earnings, attract international investment, increase tax revenues, and create new jobs has served as incentive for developing countries to promote tourism as an engine of growth.

In recent years, tourism's role in poverty alleviation has been slowly recognized among development agencies. "Pro-poor" tourism development strategies are geared toward generating net benefits for the poor. However, properly designing and implementing tourism projects remain generally a difficult process. There are both theoretical and practical challenges in justifying public sector investments in tourism and properly measuring the projects' benefits and sustainability. There is a need to come up with an analytical framework that would address these challenges and help evaluate a tourism project's outcomes. This technical note introduces a simple framework that would underpin the systematic economic analysis of tourism projects.

The structure of this paper is as follows. The rest of this section introduces the rise of propor tourism in recent years and the Asian Development Bank's (ADB) tourism projects. Section II proposes an analytical framework for economic analysis of tourism projects. Sections III to VI detail each step of the proposed framework. Section III establishes the economic rationale of tourism projects and Section IV discusses demand analysis using tourist profile and expenditure data. Section V outlines two alternative approaches to estimating tourism benefits. Section VI looks at distribution and sustainability issues. Section VII illustrates how these approaches could be put into practice using the Sambor Prey Kuk Cultural Tourism Development Project in Cambodia as an example. Finally, Section VIII provides the conclusions.

A. The Rise of Pro-Poor Tourism

Asia has become the fastest growing tourism market in the world for the past century (WTO 2006). Worldwide, the number of international arrivals shows a huge increase from 25 million in 1950 to an estimated 763 million in 2004, corresponding to an average annual growth rate of 6.5%. During the same period, the number of international arrivals in Asia and the Pacific region grew from 0.2 million to 153 million. This corresponds to an average annual growth rate of 13%, exceeding that of the Middle East (10%), Europe (6%), and Americas (5%). International tourism receipts in 2003 represented approximately 6% of worldwide exports of goods and services and nearly 30% of service exports (WTO 2006). In Asia and the Pacific region, international tourism receipts increased from \$40 million in 1950 to \$125 billion in 2004. People's Republic of China (PRC), Malaysia, Thailand, Japan, Republic of Korea (henceforth Korea), Indonesia, and India are the major tourism destinations in Asia (Figure 1).

TOURISM FOR PRO-POOR AND SUSTAINABLE GROWTH: ECONOMIC ANALYSIS OF TOURISM PROJECTS TUN LIN AND FRANKLIN D. DE GUZMAN

41,761 25,739 15,703 13,655 11,737 7 10,034 8,324 7,479 6,138 11,269 8,999 8,198 5,818^{6,053} 5,321 Republic of Indonesia India Macao, Japan **PRC** Malaysia Hong Kong, Thailand China Korea China ■ International arrivals ('000) ■ International tourism receipts (\$ million)

FIGURE 1
TOP ASIAN TOURISM MARKET BY INTERNATIONAL TOURIST ARRIVALS AND TOURISM RECEIPTS, 2004

Source: World Tourism Organization (2006).

Asia's tourism sector still has considerable room for growth for two reasons. First, the continent is endowed with a rich cultural and natural heritage that is currently not fully exploited (Harrison 1992). Taking the United Nations Educational, Scientific and Cultural Organization designated world heritage sites as an indicator of tourism resource endowments, Europe and North America emerge with 436 sites (out of the world total of 830 sites) and an annual international tourist arrival of 509 million. In comparison, Asia has 167 world heritage sites and an annual international tourist arrival of 153 million. In terms of tourist arrivals per site, Asia is still lagging behind Europe and North America by nearly 20%. Second, with its vast population and increasing income, Asia will have an increasing number of domestic tourists.

For a long time, tourism in developing countries has been described by some scholars as an exploitative form of "neocolonialism" (Britton 1981, Brohman 1996) or "leisure imperialism (Crick 1989), which suggests that tourism perpetuates poverty in third world countries. Unmanaged tourism development in developing countries has been criticized for its drawbacks (Table 1). In recent years, tourism's role in poverty alleviation has been slowly recognized by academics and international development agencies. Tourism is increasingly being included in national poverty reduction strategies (Ashley et al. 2000, Shah 2000). The rise of new "pro-poor" tourism development strategies explicitly seeks to maximize tourism benefits to the poor while simultaneously reducing its negative impacts. Proponents argue that pro-poor tourism is potentially more conducive to

SECTION I

poverty elimination than other economic sectors (DFID 1999, Ashley et al. 2000 and 2001b, Roe and Urquhart Khanya 2001).

While the concept of pro-poor tourism is relatively new, the idea of an approach to providing opportunities for the poor is sound. Tourism is pro-poor if it provides: (i) economic gain; (ii) employment; (iii) opportunities for small and medium-size enterprises (SME); (iv) infrastructure such as improved access to potable water, communications, roads (access to markets), and improved health and education services; (v) protection of natural and cultural resources; and (vi) opportunities and capacity for the poor to improve their livelihoods (ADB 2005).

Economic gain for the poor is generated since tourism brings the consumers to the natural resources and cultural heritage sites. These areas are usually occupied by poor communities. This gives the poor access to markets in which to sell their goods and services. Tourism also has the potential to generate pro-poor employment because it is a highly labor-intensive industry. The industry usually does not require high levels of skill and is thus able to create employment relatively quickly. It could also generate incomes because its investment cycle is shorter than some other industries—it only requires packaging of existing attractions and services, filling out missing components, and introducing proper management practices (ADB 2005). It also appears to offer more job and income-earning opportunities to women than do other sectors (Hemmati 1999).

Opportunities for SMEs arise because tourism development requires substantial inputs from financial services, building materials, construction, transportation, telecommunications, agriculture, food processing, arts and crafts, and other sectors of local economies. Well-planned tourism makes a powerful contribution to the formation and strengthening of linkages with other sectors of the economy and, in the process, stimulates the creation and expansion of micro enterprises and SMEs at the local level, and helps to generate supplementary cash income opportunities for women and other disadvantaged groups (ADB 2005).

In addition to employment and business opportunities, tourism development often brings with it improved infrastructure, security, communications, community development, and bolstered local pride (Roe et al. 2002). Health and education services for the poor can also be improved since government revenues generated by tourism activities may subsequently be invested in these services, as suggested by Goodwin (2000).

Tourism—particularly nature-oriented or culturally oriented tourism—often depends on the assets or natural capital of the poor (Roe and Urquhart Khanya 2001). Thus, tourism development that incorporates the protection of natural and cultural resources is pro-poor because it takes care of such assets. Moreover, this ownership by the local people of unique tourism resources provides them with opportunities to leverage their assets to obtain equity in joint venture partnerships, as well as to extract value and decision-making power.

Table 1 shows the stark differences between unmanaged tourism and sustainable and propoor tourism.

TABLE 1
UNMANAGED TOURISM DEVELOPMENT VS. SUSTAINABLE AND PRO-POOR TOURISM DEVELOPMENT

UNMANAGED TOURISM DEVELOPMENT

- Concentrated in a handful of destinations
- Weak backward linkages with other sectors (especially agriculture) in the economy and low capability to generate local income and jobs
- Erodes the value of important natural and cultural heritage assets
- Causes the loss of urban environmental amenity
- Contributes to undesirable social impacts including exploitation of vulnerable groups notably women, children, and ethnic minorities

SUSTAINABLE AND PRO-POOR TOURISM DEVELOPMENT

- · Brings the consumers to places where the poor reside
- Strong backward linkages with other sectors and good capability to generate local jobs and income
- Improved management of natural resources or cultural assets owned by the poor
- Strengthens the vulnerable group's decision-making power through their ownership of unique tourism
- Improved infrastructure, security, communications, community development, and local pride
- Indirectly finances health and education through improved government revenues

As DFID (1999) explains, the potential of tourism in contributing to poverty alleviation is compelling for the following reasons:

- (i) Tourism is a massive and growing industry already affecting millions of poor, so a marginal improvement could generate substantial benefits.
- (ii) Tourism has pro-poor growth advantages over other sectors in relation to poverty elimination.
- (iii) Although the poorest may not benefit directly, the costs they face can be reduced. Benefits can be expanded for the fairly poor such as street vendors, and casual and unskilled workers and artisans, which may also indirectly benefit their poorer relatives and neighbors.
- (iv) Considerable progress in placing environmental issues on the tourism agenda shows that concerted action can make a difference.
- (v) The limited evidence available suggests that a range of strategies can be used to develop pro-poor tourism. Little is being done in practice and so much remains to be done.

B. ADB Tourism Projects

To date, the Asian Development Bank (ADB) has approved 22 technical assistance grants (nine of which are regional technical assistance projects or RETAs) totaling about US\$8.8 million; and five loans (for three investment projects) amounting to US\$62.6 million to support tourism development (Appendix 1). The first investment project, NEP: Tourism Infrastructure Development, consisted of a package of modest but strategic components to enhance private sector activity by providing tourism-related infrastructure and integrating the tourism sector and the local economy through community participation, training, and environmental improvements. The succeeding project, NEP: 2nd Tourism Infrastructure Development Project, was designed to blend critically needed infrastructure improvements with extension of ecotourism development and environmental protection of key tourist destinations. The third project, GMS: Mekong Tourism Development Project, targeted to improve high-priority, tourism-related infrastructure in the lower Mekong basin;

Section II An Analytical Framework and Theoretical and Practical Challenges

promote pro-poor, community-based sustainable tourism in rural areas; and strengthen subregional cooperation. ADB has helped the six Greater Mekong Subregion (GMS) countries and the South Asia Sub-regional Economic Cooperation (SASEC) countries pool their unique ethnic, cultural, and natural attractions for maximum mutual benefits. It has provided technical assistance for capacity building and skills development, master plans, tourism planning, and fora.

ADB's Medium Term Strategy II aims to promote operations that are geared toward productivity-enhancing activities and employment-intensive rural development. Tourism projects and related activities that promote inclusive growth could have important roles to play.

II. AN ANALYTICAL FRAMEWORK AND THEORETICAL AND PRACTICAL CHALLENGES

A. An Analytical Framework for Economic Analysis of Tourism Projects

This section proposes a simple analytical framework for the economic analysis of tourism projects. It involves four steps as shown in Figure 2. Each step is discussed in detail in Sections IV to VII.

FIGURE 2

ANALYTICAL FRAMEWORK OF SUSTAINABLE TOURISM PROJECTS



Step 1: Establishing Economic Rationale for Public Investment

The economic rationale for public investment needs to be carefully established. Depending on the type of tourism projects under investigation, different arguments about market failure and pro-poor growth initiatives can be put forward. In infrastructure-type projects such as road construction, waste treatment, and border facilitation, arguments typically lie in the public goods nature of the projects. In the preservation of natural and cultural heritage types of projects, the missing markets for these sites as well as the missing markets for tourism revenue across generations might arise and call for public intervention. Sometimes, projects are proposed to demonstrate good practices and sustainable models for tourism development. The arguments lie in the positive information externalities generated through the process. The details of each of these arguments are discussed in Section IV.

5

Step 2: Demand Analysis: Estimating the Incremental Tourist Expenditure

Tourism brings benefits in many ways, but mostly from tourists' spending. Quantifying tourism benefits starts with a good *demand analysis*: how many tourists, where they come from, how long they will stay, and how much they will spend. A "with-the-project" and "without-the-project" analysis can find out the incremental tourists' spending created by the project. The details of demand analysis are discussed in Section V.

Step 3: Benefit Analysis: Using the Multiplier Approach to Assess Tourism Impacts

Most studies use the multiplier approach to assess the economic impacts of tourism instead of using conventional cost-benefit analysis. The "multiplier approach" tracks incremental tourist spending in the economy and estimates gross domestic product (GDP), income, and employment created while the money circulates within the economy before it "leaks" out. A small guest house in a remote province will have less leakage compared to a big international chain hotel whose supplies mostly come from outside of the country. The income multiplier based on direct and indirect impacts of tourism is then applied to incremental tourist spending to derive the economic benefits of the project. The benefits then can be compared against the costs of the concerned project(s). The details of impact assessment are discussed in Section VI.

Step 4: Distributional and Sustainability Analysis-Preservation Principle

Distributional analysis of pro-poor tourism projects has to show evidence that a significant portion of tourist expenditure will be spent outside the big chain hotels and airlines, and the poor and the vulnerable will actually have a share in the tourism benefits.

Standard institutional and financial sustainability analyses need to be conducted to ensure there are sufficient institutional and financial resources to maintain the flow of project benefits. When a government subsidy is involved, the fiscal impact of the project should be considered. The project's proceedings (ticket sales for example in the tourism site) need to sustain the operational and maintenance costs of the project.

In addition to the institutional and financial sustainability analyses, the project has to ensure that resources are adequate to preserve these tourists and heritage sites. Let us call this "preservation principle" to be further discussed in Section VII.

The above analytical framework is proposed as a practical approach to assess economic viability of a tourism project. However, a number of theoretical and practical challenges have to be noted in analyzing the tourism sector.

B. Theoretical Challenges

1. Missing Market: Pricing Natural and Cultural Heritage

From the economic point of view, how much to preserve these natural and cultural heritage assets today depends on how much the future generations will value their existence. However, due to the missing market across generations, a problem that economists refer to as "dynamic externalities—attaching shadow prices for these assets is extremely difficult, if possible at all. Future generations would not be able to "trade" with the current generation for these assets, so economic theories provide little guidance on how much preservation efforts are needed today (underexploitation is equally uneconomic as overexploitation). Intergenerational trade-off is the first type of trade-off to be considered for heritage preservation. Tourism can either accelerate or decelerate the rate of decay of these assets, depending on the balance between the additional preservation funds made available and the additional pressure put forward by tourism.

2. Missing Market: Development and Preservation

The second trade-off is development benefits versus preservation value. Natural and cultural heritage can be seen as consumption goods in themselves (which people derive value from their existence) as well as investment goods (from which people derive future economic benefits). If one takes the latter perspective, trade-offs have to be made between the present's tourism development benefits and the future's tourism development benefits. The future's tourism benefits are derived from today's preservation efforts. Economic theory indicates that one makes trade-offs between current income from tourism-induced economic development and future income flow from preservation. However, when facing enormous development and population pressure today, people might not choose the optimal depleting rate of tourism resources. The problem of myopia (very large discount rate for future income) may take place if local residents or governments have short time horizons.

3. Coordination Problem and Tragedy of Commons

Coordination problems may lead to the underexploitation of tourism resources. The tourism industry consists of a whole range of interrelated enterprises wherein the success of an enterprise's own decision will depend upon the decision and action of others. For example, excessive competition among related enterprises may lead to duplication of efforts and resources, leading to falling service quality. The "tragedy of the commons" problem leads to overexploitation of tourism resources when property rights are weak or missing. Communal property ownership is environmentally degrading and results in the "tragedy of the commons" because individual users exploit common property rather than consider its long-term sustainability. The use of national parks and cultural heritage sites are usually open to all. However, commercialization and increasing population steadily erode the values that visitors seek in these places. For example, most tourists that traditionally travel in large batches can bring destruction to an environment that most likely can sustain only a certain number of people. In this sense, many individuals, acting independently and in what they perceive to be their own interests, destroy a collective good.

C. Practical Challenges

1. Defining "Tourist"

There are a variety of definitions for tourists. Studies are not comparable to each other unless they use the same definition for tourist. The first step of tourism analysis therefore would be to carefully examine who the tourists are. Are they business travelers, pleasure travelers, or both? Are they international travelers, out-of-area domestic travelers, and/or within-the area domestic travelers? Are people who come to the tourist area to visit friends and relatives included?

2. Defining "Tourism Sector"

Another challenge is that the tourism industry usually is not defined separately in a country's industrial classification system. For example, the Standard Industrial Classification (SIC) system does not have a separate tourism sector. Instead, tourism is generally embedded in a number of different industry sectors, and each sector has both tourist sales and nontourist sales. In the SIC system, tourism-directly-related sectors include "eating and drinking" (SIC 58); "hotel and lodging" (SIC 79); "amusements and recreation services" (SIC 79); some of "retail" (SIC53, 54, 59); and "transportation" (SIC 41, 42, 44, 49 etc). In each of these sectors, there are both tourist and nontourist sales. Therefore, disaggregation of the sectors into sales attributed to tourism and sales attributed to nontourism is necessary.

3. Tourism Data

In developing countries, consistent and accurate data on the tourism industry are scarce. This lack of information is perhaps one of the industry's greatest barriers to reaching its potential. In fact, in these countries, collecting primary data for the tourism sector is vital to the development and implementation of a comprehensive and strategic marketing and public investment plan. The public/private decision makers need information on the characteristics of the tourism market to make quality marketing and investment decisions.

III. ESTABLISHING PROJECT RATIONALE FOR PUBLIC SECTOR INTERVENTION — MARKET FAILURE ARGUMENTS AND PRO-POOR PRINCIPLE

There is generally limited direct support to tourism by the public sector since it is basically perceived as a private sector activity. However, there has been a rethinking on the public sector's role and how it could effectively support the private sector for it to be able to take advantage of the potentials of tourism. The public sector can support and promote the effective involvement of the private sector in tourism by providing basic infrastructure facilities, developing human resources for tourism, increasing the capacity of public sector institutions to interact better with the private sector, and improving the linkages of tourism activities with other sectors. In particular, the government has dual roles to play in tourism development. The public sector develops and rehabilitates infrastructure to spur tourism while at the same time protects the environment and cultural/natural heritage sites.

A. Building Infrastructure—Coordination Failure and Public Goods

The participation of the private sector in tourism planning, investment and development, promotion, remains beset with a number of obstacles such as weak infrastructure, economies of scale, and high promotion and information costs, among other things. Thus, the public sector role in tourism development generally consists of setting up the infrastructure foundation (e.g., road construction, waste treatment, border facilitation) and providing the regulatory framework within which the private sector can operate.

A government targets its involvement by minimizing physical constraints and infrastructure bottlenecks (e.g., improving surface/air transport) that hamper tourism development and the movement of tourists across tourism sites.

Natural and cultural heritage assets that are of importance for tourism are threatened by inappropriate use. The quality standards and standards of hygiene could be low, especially among small-scale ancillary enterprises in the more remote tourism sites, making it difficult to attract more tourists to these areas. Government could assist in improving the environment—a public good, or a collective good—that would not be provided by individuals or private sectors without coordination.

A similar argument applies for efforts to collectively promote tourism destinations through chambers of commerce, group advertisement, national image building, etc. The public sector can play an important role in these activities.

Government could also try to minimize institutional constraints such as visa procedures, repatriation of foreign investment proceeds, etc. At the same time, it implements measures aimed at encouraging private sector participation in other areas including investment, marketing, and technical training. This is done by providing the regulatory framework such as approval of accommodation establishments, issuance of licenses to travel and other adventure-related activities, and promotion of tourism in and outside the country.

B. Preserving Natural and Cultural Heritage

Natural and cultural heritage resources are the primary resources upon which tourism depends, now and in the future. These include not only the important natural and cultural sites but also the intangible cultural heritage as represented by the unique customs, traditions, and arts and crafts especially of ethnic minorities living in the vicinity of these sites. Investments are needed for access and protection of infrastructure in natural, cultural, and urban sites to strengthen the sustainable management of tourism. Investments in improved heritage management practices; a more inclusive approach to tourism development to accelerate poverty reduction and reduce undesirable social impacts; and enhanced quality of services and investments in the ancillary sector are also needed.

C. Demonstrating Good Practices and Sustainable Models—Information Externality

The public sector is also involved in providing demonstration models to promote and manage sustainable tourism at heritage and urban sites. A demonstration model shows the approach that

should be taken in developing pro-poor tourism initiatives in pilot studies. The model aims to: (i) increase substantially the volume of poverty reduction generated through tourism; (ii) help secure and conserve key cultural and natural tourism assets through alternative livelihood development; (iii) ensure that women and ethnic minorities are given equitable opportunities to participate in and benefit from tourism economy; and (iv) provide a body of knowledge and experience on poverty reduction through tourism that can be replicated elsewhere.

IV. DEMAND ANALYSIS: TOURIST PROFILE AND EXPENDITURES

Typically, the demand analysis begins with a tourist/visitor profile. The profile can be obtained through government or industry records or, most commonly, through a visitor survey. The survey asks questions like who the tourists are, where they come from, what they do in the region, etc.

Forecasting tourism demand is essential to the tourism project analysis. Unrecoverable losses due to high vacancy rates will result if demand forecasts are unduly optimistic, and vice versa (Uysal and Crompton 1985 in Calatone, et. al. 1987). Tourism demand is typically measured by the arrival and expenditure of tourists by areas of origin and destination, types of attraction and accommodation, length of stay, modes of transportation, and kinds of promotion, etc.

Tourism forecasts may be generated by either qualitative or quantitative approaches. The main focus of qualitative approach is the Delphi method (Linstone and Turoff 1975, Rowe and Wright 1993, Weaver 1972, Woudenberg 1991), which uses a series of iterative survey questionnaires to generate a general consensus among a group of experts.

Quantitative methods that have been applied to tourism demand include econometric and time series models. Econometric models estimate the causal relationship between tourism demand and its determinants using regressions to fit a linear or log-linear model with a set of explanatory variables. Major explanatory variables usually include exchange rates, income in the country of origin, and relative price levels etc. This methodology has the advantage of simplicity, but it also has several serious limitations including model misspecification, different error distribution, and data availability. Moreover, some have doubted the general validity of estimating a specified demand model in terms of ex post explanatory variables, as it relies on the assumption that the explanatory variables will determine the tourism demand in the future, in the same pattern, trend and magnitude as in the past.

On the other hand, the time series forecasting models extrapolate historical trends of tourism demand into the future without considering the underlining causes on trends. Typical time-series methods include constant growth, exponential smoothing, univariate Box-Jenkins, autoregression, moving averages, and decomposition. Spatial models have also been used in tourism forecasting (Petropoulos et. al. 2005, Witt and Witt 1995). Compared to the econometric models, these models have less data requirements, but they are not based on any theory that underlies tourist's decision making process.

For the purpose of project evaluation, changes of tourist arrival and profile need to be estimated under both "with" and "without-the-project" scenarios. This is difficult as it heavily relies on counterfactual assumptions on the marketing and sales of various subsectors of tourism. A careful analysis is needed to evaluate the historical trends of tourist profiles as well as the site's carrying capacity. In many cases, it is assumed that, under the "without-the-project" scenario,

SECTION IV DEMAND ANALYSIS: TOURIST PROFILE AND EXPENDITURES

tourist arrivals will continue its historical trends until reaching the site's existing capacity, while tourist arrivals increase under the "with" project scenario. The incremental tourist arrivals, as well as their expenditures, are attributed to the project.

Based on the estimated tourism profiles and arrivals, incremental tourist expenditures are calculated, where "incremental" means the difference in total tourists' expenditures between "with" and "without-the-project" scenarios. The calculation of total tourist expenditures takes one of two approaches: demand-related (expenditures reported by tourists) or supply-related (receipts received by businesses from tourists).

incremental tourist expenditure = tourist expenditure with the project - tourist expenditure without the project (1)

For the demand-related approach of estimating expenditure, tourists are categorized by type such as day travelers, campers, overnight hotel guests, etc. The average expenditures and the number of tourists of each type are estimated. Total tourist expenditures are derived by multiplying the number of tourists with the average expenditures by type and summing over the types.

total tourist expenditures =
$$\sum_{i=1}^{n}$$
 (no. of tourist of type_i)(average expenditures per visit_i) (2) where i =tourist types

Typically, such a demand-related approach relies on constructing tourist profiles through tourist registration records or tourist surveys. Various models have been developed to standardize the procedures for constructing tourist profiles and examining different types of tourists and their expenditure patterns. One example is the Travel Economic Impact Model (TEIM) model used by the US Travel Data Center to estimate tourist expenditures. However, the difficulty mostly lies in the data. Tourist numbers estimated from airport arrivals, welcome center registration, park-gate interviews, and traveler surveys can be inconsistent across places and subject to errors. Especially for some tourist types such as day travelers and local visitors, registration records are usually neither well kept nor consistent across places.

For the supply-related approach (receipts received) to calculate tourists' expenditures, industry groups that serve tourists (lodging, eating and drinking, recreation etc.) have to be identified along with the average sales to tourists by each industry group. Tourist expenditure is estimated by multiplying the numbers of businesses within a group by average sales to tourists and summing over groups.

tourist expenditure =
$$\sum_{i=1}^{industry\ groups}$$
 (no. of firms in group_i) (average sales to tourists_i) (3) where $i=industry\ groups$

One of the advantages of the receipts received approach is that it is usually easier to gather information from businesses that are in the study area and that usually keep business records. A shortcoming with this approach is the percentage of goods and services in these industry sectors sold to tourists is difficult to estimate and often subject to substantial errors. It is easier for some

TOURISM FOR PRO-POOR AND SUSTAINABLE GROWTH: ECONOMIC ANALYSIS OF TOURISM PROJECTS TUN LIN AND FRANKLIN D. DE GUZMAN

sectors such as lodging to estimate percentage of sales from tourists than for other sectors such as eating and drinking or retail sectors.

There are also many studies that combine supply-related and demand-related approaches to derive tourist numbers and subsequently expenditures. For example, Longwoods International Inc. in their tourism study used a supply-related model named TRAITS II (Tourism Analysis and Information Tracking System) to calculate total hotel/motel/resort and camping rental revenues. A demand-related travel spending behavior model was used to project total tourists' expenditures by category. Using the estimated tourists' expenditures by category and the hotel/motel/resort revenue information, the studies were able to estimate the number of tourists and their total expenditures.

There are two advantages of combining demand-related and supply-related approaches to estimate number of tourists and their expenditure. First, supply-related approaches, which usually include conducting a lodging survey, can estimate reasonably accurate lodging business revenues from tourists because presumably the lodging sector keeps good occupancy records. Second, demandrelated approaches, which usually include conducting a tourist survey, can estimate reasonably accurate tourist expenditures by category. Combining lodging revenue and the percentage of tourist expenditure on lodging, one is able to estimate total tourist expenditure.

There is one last point to be noted-demand analysis is based on certain assumptions about the "substitution effect." It is very important to consider, when visitors incur additional spending in the tourism sites either by staying longer or spending more per unit of time, whether or not this spending "crowds out" their spending elsewhere. If the answer is yes, the substitution effects need to be considered. In this regard, it is usually helpful to separate domestic visitors from foreign visitors when conducting the demand analysis. It is obviously easier to ignore the "substitution effects" of foreign visitors' spending, which might take place at their home country, than to ignore the "substitution effects" of domestic visitors' spending, which could have taken place somewhere else within the concerned economy. As a matter of fact, the substitution effects of domestic visitors' spending could be substantial (as much as 100%) depending on, without tourism, how much of domestic visitors' money would have been spent abroad by importing foreign goods and services. Otherwise, if the money would have been spent entirely within the economy, tourism does not generate real "incremental" spending. This is the reason that in many national tourism impact studies, only foreign visitors' spending is used for calculating the economic impacts. Domestic visitors' spending yields impacts only when at least one of the two following conditions is fulfilled:

- without the project, (some of) domestic visitors' spending would have been spent on importing foreign goods and/or services
- (ii) the tourism sector has "greater linkages" when the tourism sector buys a greater share of its inputs from domestic sectors (larger "multiplier")

This will be further illustrated in Section VI using a real world example.

SECTION V ESTIMATING TOURISM BENEFITS: TWO ALTERNATIVES

V. ESTIMATING TOURISM BENEFITS: TWO ALTERNATIVES

Estimating Benefits: Economic Impact Analysis versus Cost-Benefit Analysis A.

Different approaches have been applied to estimate the economic benefits of the tourism industry. Two most commonly used approaches are (i) economic impact analysis based on multipliers, and (ii) cost-benefit analysis based on evaluation of nonmarket goods. These two approaches focus on different aspects, which need to be clearly understood.

An economic impact analysis based on multipliers traces the flows of spending associated with tourism activity in a region to identify changes in sales, tax revenues, income, and jobs due to tourism. The principal methods here are visitor spending surveys, analysis of secondary data from government economic statistics, input-output models, and multipliers (Frechtling 1994). Multipliers are the impacts generated in a tourist's destination for every dollar that is spent on the tourism activity (Tooman 1997). Multiplier analysis is a tool to estimate indirect impacts from direct impacts. Generally, multipliers are derived from the total impact of changing one variable on other related variables in an economy (Tanjuakio, Hasting, and Tytus 1996). Depending on the variables under consideration, one can estimate output multipliers, employment multipliers, income multipliers, and so forth.

There are different techniques to estimate multipliers such as standard multiplier analysis, econometric analysis, input-output analysis, social accounting matrix (SAM) analysis, and computable general equilibrium model (CGE). Each is discussed in the following sections.

A cost-benefit analysis based on valuing nonmarket goods and services estimates the relative economic efficiency of alternative policies by comparing benefits and costs over time from the perspective of societal welfare. Many tourism assets are nonmarket goods (e.g., national parks, wetland conservation, etc.) and the demand and value of these nonmarket goods arise from both actual use and nonuse of these goods. Benefits are derived from direct use of the resource such as recreational experience from visiting a park, hunting, fishing, boating, camping, or enjoying a panoramic view. These are measured by the consumer's surplus in the Hicksian sense. The questions asked are (i) "how much would recipients be willing to pay to gain access to the recreational resources under consideration?" or (ii) "how much would they be willing to be paid to abstain from the opportunity?" Nonuse benefits arise not from present use but from the option of future use or knowledge of continued existence of the resource. This is also referred to as preservation value. It has various forms including option, existence, and bequest value.

Cost and benefit analysis makes use of a wide range of methods for estimating values of nonmarket goods and services, such as direct market pricing, surrogate market pricing, costanalysis, travel cost method, contingent valuation, conjoint choice experiment, hedonic pricing, etc. The travel cost method, for example, estimates consumer's surplus based on their varying travel costs. Contingent valuation involves direct questioning of respondents and obtaining their willingness-to-pay or willingness-to-accept data (Stokey and Zeckhauser 1978, Sudgen and Williams 1978, Pabuayon 1991).

TOURISM FOR PRO-POOR AND SUSTAINABLE GROWTH: ECONOMIC ANALYSIS OF TOURISM PROJECTS TUN LIN AND FRANKLIN D. DE GUZMAN

These two approaches (economic impact analyses based on multipliers, and cost-benefit analysis based on evaluating nonmarket goods) are frequently confused since both discuss economic "benefits." There are two clear distinctions between the two approaches in that they measure different things with different perspectives:

- What Benefits? The cost-benefit analysis includes market and nonmarket values (consumer surplus), while economic impact analysis is restricted to actual flows of money from market transactions. In the cost-benefit analysis framework, all nonmarket benefits (e.g., the existence value or bequest value) and social costs (e.g., increased trafficking in women and children, child labor, increased HIV prevalence due to increasing tourist numbers etc.) need to be quantified. In the economic impact analysis, only market benefits and costs are considered and nonmarket benefits and social costs need to be discussed separately.
- (ii) Whose Benefits? The economic impact analysis based on multipliers focuses on the regional distribution of economic activity. The income received from tourism by a destination region is largely offset by corresponding losses in the origin regions, yielding only modest contributions to net social welfare and efficiency. However, the cost-benefit analysis addresses the welfare changes of all the beneficiaries, including foreign visitors and future generations.

In this paper, economic impact analysis is introduced based on multipliers for three reasons. First and most importantly, usually ADB public investment projects take a national perspective, comparing project economic benefits with costs at the national level. A developing country's investment on tourism sites should not be justified based on the welfare changes (travel costs or willingness-to-pay for example) of visitors, because many of the visitors are from abroad. As a matter of fact, the investment should be questioned if its benefits mainly accrue to foreign visitors, which is exactly one of the reasons that tourism in developing countries has been conceptualized as an exploitative form of "neocolonialism" or "leisure imperialism." Second, economic impact analysis based on multipliers has been used throughout previous ADB tourism projects and therefore, the approach deserves some careful examination.

Before proceeding to discuss economic impact analysis based on multipliers, the approach's limitations need to be noted. First, it needs to be understood that the "benefits" derived from the multiplier analysis are the changes in economic activities (gross domestic product, jobs etc.) within a geographic region rather than changes in consumer surplus or welfare. As earlier stressed, income received from tourism by a destination region is largely offset by corresponding losses in the origin regions. Hence, the economic benefits measured by this approach are not welfare changes. Second, multiplier estimates are heavily influenced by the size of the study area. Generally, the larger the study area, the higher the multiplier. Third, unlike life-cycle analysis for example, most multiplier techniques provide static state analyses only. Approaches like life-cycle analysis analyze the differential impacts of evolutionary stages of tourism development stages from exploration, involvement, development, consolidation, stagnation, to decline/rejuvenation. See Box 1 for types of multipliers.

Box-1 Multipliers

Generally, economic multipliers estimate the economywide impact on related variables from changing one variable in the specified economy (Tanjuakio, Hasting, and Tytus 1996).

- Output Multiplier: An output multiplier for a sector is defined as the total production in all sectors of the economy that is necessary to satisfy a dollar's worth of final demand for that sector's output (Miller and Blair 1985).
- Personal Income Multiplier: Every dollar change in final-demand spending (direct output) results in a change in income received by households.
- Employment Multiplier: Every dollar change in final-demand spending (direct output) in a sector results in a change in number of jobs in the economy.
- Business Taxes Multiplier: Every dollar change in final-demand spending (direct output) results in a change in business taxes (e.g., sales taxes).

Each of the above multiplier effects can be further disaggregated into direct effects, indirect effects, and induced effects.

- Direct Effect: the initial impact on an economy; production changes associated with changes in demand for the good itself
- Indirect Effect: the secondary impact caused by changing input needs of directly affected industries (e.g., additional input purchases to produce additional output)
- Induced Effect: caused by changes in household spending due to the additional employment generated by direct and indirect effects.

An example best illustrates these concepts. Consider the income multipliers for tourists' spending in Region X to be 0.3, 0.1, and 0.2 for direct, indirect, and induced multipliers, respectively. That means, for every one dollar tourists spend in Region X, 30 cents will be added to tourism sector income. Moreover, 10 cents will be added to all other related sectors' income when the tourism sector buys inputs from sectors, and these sectors buy inputs from more sectors. Another 20 cents income will be generated due to the consumption by newly created employment. Depending on the purpose of the calculation, the total multiplier either equals the sum of direct and indirect multipliers (0.4); or the sum of direct, indirect, and induced multipliers (0.6). This means that for every dollar brought in by tourists, 60 cents income are generated in Region X.

B. Standard Multiplier Analysis

The standard multiplier approach is the simplest and least expensive way to calculate multipliers (Leitch and Leistritz 1985). The formula for calculating the standard income multiplier is:

Income Multiplier =
$$\frac{1}{1 - (MPC \times PSY)}$$
 (4)

The marginal propensity to consume locally (MPC) represents the proportion of local income spent within the locality. One can make a crude estimate of the MPC by asking "what proportion of people's incomes is likely to be spent locally?" Empirical estimates vary from 0.20 to 0.80. PSY is the proportion of a tourist expenditure dollar that becomes income to local households, which typically ranges from 0.25 to 0.75. This variable measures how much local labor, interest, and profit is involved in the final price of the product. For example, one would expect to see a higher PSY in locally operated tourism businesses that hire mostly local people, than international chains that employ more nonlocal residents (Blakely 1994).

TOURISM FOR PRO-POOR AND SUSTAINABLE GROWTH: ECONOMIC ANALYSIS OF TOURISM PROJECTS TUN LIN AND FRANKLIN D. DE GUZMAN

Another method to estimate the first approximation of a standard multiplier is known as "economic/export base analysis" (Leistritz and Murdock 1981, Oppenheim 1980, Bendavid-Val 1983). The equation for this method is:

$$Multiplier = \frac{\text{total Employment / Income / Activity}}{\text{export Employment / Income / Activity}}$$
(5)

For example, if total employment in a region were 3,000 of which 1,000 employees worked in the export sector, then the employment multiplier would be:

Employment Multiplier =
$$\frac{3,000}{1,000}$$
 = 3 (6)

The standard multiplier approach has been used in tourism studies since the 1960s. The early works by Zinder (1969) estimated a high tourism sector multiplier relative to other industrial sectors. Further empirical work showed that the tourism industry did not provide the vigorous growth effect suggested by these early studies. While these studies were being scrutinized, the opinion that the tourism industry has a greater than average multiplier was still held.

The standard multiplier approach has been criticized for its oversimplification. However, no other model can approach the export base techniques (standard multiplier approach) in terms of inexpensiveness and simplicity (Pfister 1976).

Econometric Techniques C.

These techniques involve statistically estimated equations that explain factors determining economic growth. Most common techniques used are regression models evaluating the relationship between tourism and economic growth. The primary difference between using econometric models and using standard multiplier approaches to estimate economic impact is temporal. Standard multiplier models use one single period reference whereas econometric models employ time-series data (Pleeter 1980).

Feder (1983) and Ram (1986) developed an econometric model to evaluate the impact of tourism on the economy. The study includes tourism output growth as a determinant of national income growth. Modeste (1995) examined some Caribbean countries using the same model specification and the results showed a statistically significant positive relationship between growth of tourism output and growth in income.

The use of econometric models to estimate regional economic impacts due to tourism has several advantages. Econometric models can accommodate a vast array of determinants such as capital stock. Another advantage in using these models is the sense of accuracy they impute and their ability to establish confidence limits on the estimates, which is lacking in other techniques such as standard multiplier analysis.

Two serious shortcomings exist when using econometric models to estimate impacts of tourism. One is data limitation and the other is model specification. When many variables are involved with limited time-series data, estimation becomes problematic due to the small degrees of freedom.

Section V Estimating Tourism Benefits: Two Alternatives

While both time-series and cross-sectional data exist at the national level, similar data at regional levels are scarce. Model specification problems involve a number of issues, from data availability and reliability to properly applying statistical and economic theory.

D. Input-Output Analysis

An input—output (I/O) analysis uses an economic model that traces the flow of goods and services, income, and employment among related sectors of the economy. The I/O approach triggers the flow of activities as follows: When final demand for a good changes, the sector producing the good (output) purchases inputs from other industrial sectors, which in turn purchase inputs from other industries. Moreover, all of these industrial sectors purchase additional labor input. The employees use their compensation to purchase goods and services from the economy. Linkages among industries in a region create a ripple effect as a result of change in demand for a product. Strong linkages can lead to healthier economies, as capital flows through the economy rather than out of it. An I/O model is based on the premise that the economy can be decomposed into aggregate sectors. The I/O model is therefore a tabular representation of output flows from each of several industries or sectors and the flows of inputs to various industries or sectors.

The idea of an I/O model is simple. In an economy, sector output is denoted by vector X and vector Y is the final demand in each sector. They are connected through a matrix $(I-A)^{-1}$, known as "Leontief inverse", as shown in the following equation:

$$X = (I - A)^{-1} \bullet Y \tag{7}$$

The inverse matrix $(I-A)^{-1}$ is at the heart of an I/O analysis, where I is the identity matrix and A is the I/O coefficient matrix across sectors. An I/O model gives the I/O coefficient matrix A, so matrix $(I-A)^{-1}$ is derived. The change of outputs (ΔX) , therefore, can be estimated from any change in final demand (ΔY) as illustrated in equation (9):

$$\Delta X = (I - A)^{-1} \Delta Y \tag{8}$$

In this case, ΔY is the incremental tourists' expenditure. Economic impacts of the incremental tourists' expenditure are represented in the changes of outputs across sectors (ΔX). Comparing the total change of outputs to the incremental tourists' expenditure, one will get the tourism output multiplier. Impacts on income, employment, and tax revenues, as well as their multipliers, can therefore be derived through their relations with the outputs.

Input-output analysis is the most widely used method for estimating economic impacts in terms of multiplier effects. Examples of I/O analysis in tourism studies include those by Curry (1986); Liu and Var (1982); Liu (1986); Seow (1981); Johnson, Obermiller, and Radtke (1989); Heng and Low (1990); Bergstrom, Cordell, Ashley, and Watson (1990); Dawson, Blahna, and Keith (1993); Douglas and Harpman (1995); Archer (1995); Archer and Fletcher (1996); Longwoods International (1992, 1993, 1995); and Institute for Tourism and Recreation Research (1996). Among the US states that have used I/O analysis for their tourism studies are Connecticut, Montana, Kentucky, New Jersey, New Hampshire, Pennsylvania, Vermont, and Wisconsin.

Aside from domestic studies, there are a number of international tourism studies that have used I/O analysis. Armstrong et al. (1974) studied the tourism impact on the Barbados economy using a 13 sector I/O transaction table. An output multiplier of 1.4 was obtained. Diamond (1976) used I/O to estimate the various output multipliers for Turkey; they ranged between 2.1 and 3.2. Heng and Low also employed the I/O model to analyze the economic impact of Singapore's tourism industry. They found that tourism's employment multiplier was three times that of total exports and twice that of manufacturing exports in Singapore. Archer (1995) compared the results of three separate I/O studies that measure the economic contribution of tourism on the Bermuda economy. He concluded that the level of employment in the economy depends heavily on tourism although the leading generator of foreign currency and income had been international business and finance since the early 1990s.

E. SAM Analysis and CGE Model

Social accounting matrix (SAM) has recently been adopted to calculate tourism multipliers. In Wagner's recent study on Guaraqueaba, a region in the northeastern state of Paraná in Brazil, a SAM was used to examine the economic effects of tourism in that region. A SAM is an extension of an I/O table, with institutional sectors (households, government, and capital) and factor sectors (land, labor) included. Thus, not only are production linkages considered, but also consumption, income distribution, and saving/investment. In general, SAM describes the structure of an economy in terms of the links between production, income distribution, and demand within a region's economy (Thorbecke 1985).

Use of the computable general equilibrium (CGE) framework in modeling tourism effects became popular in the late 1980s (Hanson et al. 1990). A CGE model uses SAM as its data sources. Compared to SAM or I/O approaches, a CGE model better incorporates the price effect and input substitution in production in generating the economic impacts. Zhou, Yanagida, Chakravorty, and Leung (1997) conducted a study to compare the economic impact of tourism in Hawaii using the I/O model versus the CGE model. The study concluded that the results of the I/O model are similar in magnitude to those of the CGE model but generally higher, and that sectors closely associated with tourism exhibit the largest effects.

The I/O model, SAM, and CGE all provide impact estimates in a general equilibrium framework instead of single-market analysis (referred to as "partial equilibrium"). They capture not only the direct impact of tourists' expenditure but also the indirect and induced impacts that occur when tourist dollars work their way through the economy. Therefore, they are ideally suited to measure both the relative sizes of sectors that make up the economy and the linkages among them. They produce a structural model that illuminates the interactions among many sectors, and measures impacts as they reverberate through the economy. Understanding which types of economic activities generate higher returns can direct decisionmakers toward enterprises that will stimulate economic development within the region. The increasing use of these approaches in recent decades offers another advantage in these approaches, producing more comparable results across studies.

SECTION VI DISTRIBUTIONAL ANALYSIS AND SUSTAINABILITY ANALYSIS

In practice, one of the major differences among the I/O models, SAMs, and CGEs is data requirement. The data requirements are less for an I/O model than for SAM or CGE. Constructing an I/O model is the first step in building a CGE model. Compared to I/O models, SAM and CGE models are neither quickly accomplished nor easily replicable due to the nature of the technique and intensive data requirements.

VI. DISTRIBUTIONAL ANALYSIS AND SUSTAINABILITY ANALYSIS

Distributional Analysis A.

"Unmanaged tourism" is criticized for concentrating their benefits within a few destinations and big chains, while pro-poor tourism projects are expected to bring more benefits to remote areas, poor and vulnerable groups, and SMEs. Using the tourist profiles and the multiplier approaches, economic analysis of the pro-poor tourism projects have to show evidence that a significant portion of the tourist expenditure will be spent outside the big chain hotels and airlines, and that the poor and vulnerable will actually share in the tourism benefits.

B. Financial and Institutional Sustainability

Assessing tourism project viability also requires detailed financial and institutional sustainability analysis. For revenue-generating tourism projects, a key assessment is whether they have enough funds for investment, operation, and maintenance. Where relevant, analysis is needed on the self-financing capacity of the project-operating entity through prevailing prices or user charges. In tourism projects, user charges include entrance fees, parking fees, visitor center and gift shop revenues, donations, and concession fees for market stalls. For revenue and especially nonrevenue generating projects, the fiscal impacts of the project should be considered. Where the tourism services are funded directly through the government budget, an assessment is needed of the fiscal impact of the project arising from, for example, operation and maintenance.

Assessment of institutional sustainability focuses on identifying functions, structure, and capacities of agencies. A clear and distinct role for public or private agencies needs to be established based on systematic assessment of institutional factors that underlie market and/or institutional underperformance or failure. For example, institutional factors are important in assessing whether or not a newly developed bus tour operated by a community joint venture would be successful and sustainable.

Environmental Sustainability and Preservation Principle C.

Besides financial and institutional sustainability, tourism projects have to examine another very important aspect—environmental sustainability. Sustainable and pro-poor tourism projects start with the goal of better protecting and managing natural and cultural heritage sites while making tourism beneficial to the poor. One needs to be ensured that preservation is not sacrificed while developing tourism. From the economic analysis perspective, there is a simple way of doing this, called "preservation principle." The project has to look into the preservation inputs under

TOURISM FOR PRO-POOR AND SUSTAINABLE GROWTH: ECONOMIC ANALYSIS OF TOURISM PROJECTS TUN LIN AND FRANKLIN D. DE GUZMAN

both "with" and "without" project scenarios and ensure that the preservation inputs, both in absolute terms and per visitor terms, will increase with the project. This point of analysis goes beyond the conventional financial sustainability analysis and cost recovery, as it mainly concerns the distribution of project revenues toward preservation. For example, in a project to develop a cultural heritage site, one has to analyze how many archaeologists and staff are hired and how much of the preservation money would be made available to them without the project. The analysis has to show how these inputs would change in both absolute and per visitor terms after the project is implemented and more visitors start arriving.

VII. AN ILLUSTRATIVE EXAMPLE: SAMBOR PREY KUK CULTURAL TOURISM DEVELOPMENT PROJECT1

A. **Site Description**

Sambor Prey Kuk Cultural Landscape in Cambodia is the location of the seventh century pre-Angkorian city of Isanapura, where the Sambor Prey Kuk monuments are found. Nearly 200 archaeological sites have been discovered scattered throughout more than 20 square kilometers that comprise the cultural landscape. West of the main temple clusters is the ancient city of Isanapura, by historical accounts once home to more than 20,000 households and linked to Angkor Wat by an ancient road.

B. **Project Description**

This is one of the subprojects under the GMS Sustainable Tourism Project (TA 6279). This subproject will:

- (i) map and create an updated geographic information system for the site
- (ii) develop and implement a heritage based zoning plan and training for site managers and community leaders
- (iii) undertake minor upgrading and improvement to the site's 14 km access road
- (iv) construct and install a visitor interpretation center, facilities for presentation of site features, protection infrastructure, and up to 15 km of interpreted walking trails
- (v) build the capacity of local communities in tourism-related livelihood development and cultural heritage management
- (vi) develop and implement a monitoring program

¹ Although this example is taken from the existing ADB project (GMS: Preparing the Sustainable Tourism Development Project), some numbers presented here are hypothetical and are for illustration only.

C. Project Rationale

A.LT. LIBRAR

Development of zoning plans, construction of the access road and visitor interpretation center, and preservation of the site are public/common goods. Improving access and management of visitor flows at the site and protecting the site's tangible and intangible heritage values as a tourism asset both require collective action. This provides a strong rationale for public investment. In addition, the project will serve as a model for controlling the impact of tourism on threatened cultural heritage sites and local communities in Cambodia, and hopefully provide useful information for future similar projects in the country.

D. Demand Analysis: Visitor Profile and Expenditure

In 2005, the province where Sambor Prey Kuk is located received 225,790 tourists most of whom (96%) comprised domestic tourists. Estimates by the Provincial Tourism Office suggest that international tourists stay an average of 1.61 days and spend \$34.43 per day, while domestic tourists stay an average of 2.37 days, spending an average of \$27.92 per day. The volume of international and domestic tourism arrivals to Kampong Thom province increased by an average of 41% and 15% per year, respectively, between 2002 and 2005, reflecting a growing volume of tourists going to Siem Reap by road from Phnom Penh. It also reflects interest in exploring other ancient monuments outside the Angkor Wat temples, as well as holiday vacation purposes in the case of the domestic market. In 2005, Prasat Sambor Prey Kuk received 6,800 international tourists (about 78.5% of total international arrivals to the province) and 38,000 domestic tourists (about 17.5% of total domestic tourists to the province.) International tourists to the site increased by an annual average rate of 40.3% between 2002 (2,461 visitors) and 2005. Data on the growth of domestic tourists to the site are not available but are reported to have increased at a much slower rate of less than 10% per year. The visitor forecasts "with" and "without-the-project" are presented in Table 2. The length of stay suggested by the survey averaged 0.64 days for international and 0.71 days for domestic, which includes festivals and day picnicking.

Project surveys of international and domestic tourists in Kampong Thom asked respondents about their daily tourism expenditures that were adopted for the current economic analysis. Average daily expenditure from the survey was \$34.40 for international visitors and \$27.90 for domestic visitors. The total incremental tourist expenditure is estimated by multiplying them with the corresponding incremental visitor days.

TABLE 2 FORECAST OF VISITORS FOR PRASAT SAMBOR PREI KUK

		2012	2013	2014	2015
Number of Visitors		111			l wheth
With project					i Şarak y
International	18289	21033	24188	27816	31988
Domestic	50924	53470	56143	58950	61898
Total	69213	74502	80331	86766	93886
Without project					
International	14095	10879	8396	6480	5000
Domestic .	35647	42776	44915	47160	43329
Total	49741	53654	53310	53640	48329
Length of Stay (average)					
With project	Day	Day	Day	Day	Day
International	0.64	0.65	0.67	0.68	0.70
Domestic	0.71	0.72	0.73	0.74	0.75
Without project					
International	0.60	0.53	0.45	0.38	0.30
Domestic	0.60	0.55	0.50	0.45	0.40
Visitor Days					in Abers
With project					
International	11623 × 11623	13705	16151	19022	22392
Domestic	36079	38438	40943	43601	4642
Total	***	52144	57094	62624	6881
Without project					
International	8457	5711	3778	2430	1500
Domestic	21388	23527	22457	21222	1733
Total	29845	29238	26235	23652	1883

Application of Income Multiplier E.

The income multiplier for the tourism sector in the region is estimated at 0.60. In other words, for every tourist dollar spent in the economy, 60 cents of local income is generated, including direct effects and indirect effects. Following the application of the income multiplier of 0.60, the average benefit per international visitor is estimated at $34.40 \times 0.6 = 20.7$; and for domestic visitors is $27.90 \times 0.6 = 16.8$, based on which the economic internal rate of return is calculated to be 16.4% for the project. This is based on the assumption that there is no substitution effect at all for both international and domestic visitor spending. Such assumption, as discussed in Section IV, is highly unrealistic especially for domestic visitors. Making further assumptions:

Domestic visitors' spending has a 80% substitution effect, meaning 80% of the \$27.90 is deducted from the domestic visitors' spending elsewhere within the economy.

SECTION VII AN ILLUSTRATIVE EXAMPLE: SAMBOR PREY KUK CULTURAL TOURISM DEVELOPMENT

(ii) While the tourism income multiplier is estimated at 0.6, the "average" of the income multipliers of all consumption sectors within the economy is 0.45, meaning that each dollar domestic visitors spend elsewhere on average yields \$0.50 income within the economy.

Therefore, the average benefit per international visitor is estimated at $$34.40 \times 0.6 = 20.7 and for domestic visitors is \$27.90 x (1-80%) x $0.6 + $27.90 \times 80\%$ x $(0.6-0.45) = 6.70^2 , based on which the economic internal rate of return is calculated to be 10.1%.

F. **Distribution Analysis**

Evidence is collected through field investigations, focus group discussions, and household surveys to ensure that the poor will benefit from tourism planning and development in the area. The primary source of income in Sambor Prey Kuk is rice cultivation. Both men and women, and at times children, engage in agriculture, especially during the raining season from June to September. The household survey shows that tourism-related activities are welcomed by most to supplement their household income and "fill in spare time." The poorest groups in the area, made up of women, women heads of households, ethnic minorities, war disabled, and poor resettled people are best positioned to benefit from the expanded markets for traditional crafts, like weaving, embroidery, pottery, and wood craft; cultural performance; and increased production of agricultural food items for sale. These activities not only can be managed by the poor even though they are minimally literate or educated but also are culturally acceptable to the men.

Fees collected from tourists visiting Sambor Prey Kuk will contribute to the creation of a village development fund, which brings multiple benefits in enabling villagers to discuss real options and to make decisions about investments in their community. Sometimes they use the village fund to help out their poorest neighbors, since they know who are deserving and can monitor the results.

On the downside, the social costs induced by increasing tourist number such as increased trafficking in women and children, child labor, HIV etc. are likely to be borne disproportionately by the poor.

Sustainability Analysis G.

Financial sustainability is concerned with whether the revenue collected from the project will be sufficient to cover the operation and maintenance costs of the project. Project surveys of international and domestic tourists in Kampong Thom asked respondents to indicate their interest in visiting the site if developed in the way proposed and their willingness to pay an entrance fee. The Tourism Management Committee will collect and manage the site entrance fee for international and domestic tourists, to cover use of walking trails, shuttle bus services, and visitor center. Other revenues that the Tourism Management Committee will collect are concession fees for market stalls at the gift shop. Analysis shows these measures will guarantee adequate institutional and financial resources to operate and maintain the site.

² This involves the assumption that the domestic visitors' substituted spending, which takes place elsewhere within the economy, also has an income multiplier of 0.6.

TOURISM FOR PRO-POOR AND SUSTAINABLE GROWTH: ECONOMIC ANALYSIS OF TOURISM PROJECTS TUN LIN AND FRANKLIN D. DE GUZMAN

Environmental sustainability is concerned with whether the value of the site is eroded or preserved by increased number of tourists. In Sambor Prey Kuk, the preservation spending and efforts are reviewed, in both absolute terms and per visitor terms, with and without the project. It shows that, without the project, the site will continue employing one archaeologist and around 30 supporting staff with an annual preservation budget of around \$50,000 coming from the government budget. Considering the site currently is receiving around 31,332 visitor-days a year (comprised of 4,352 international and 26,980 domestic travelers), this is an average preservation expense of \$1.60 per visitor per day. This input might decrease over the years if the site does not generate revenue to support the preservation. With the project, the site will receive an additional \$40,000 as preservation inputs. This comes from a committed government transfer from 40% of the admission fees from international visitors. The increased preservation inputs will be spent on hiring one more archaeologist and several support staff, and undertaking reconstruction work on the monuments. By 2011, the number of visitors per day to the site is going to increase to 47,702. The preservation input will increase to \$1.87, an 18% increase compared to the current level.

VIII. CONCLUSIONS

ADB's Medium-Term Strategy II aims to enhance ADB's relevance by meeting the key development challenges facing the region in reducing poverty. One of the identified strategic priorities includes strengthening inclusiveness to enable disadvantaged groups to benefit equitably from the opportunities that development provides. A major operational implication is to selectively focus on operations that promote productivity-enhancing reforms, employment-intensive rural development, and social development. In this regard, pro-poor tourism could have an important role to play.

This paper has introduced a simple analytical framework that would underpin the systematic *ex ante* economic impacts of pro-poor tourism projects. This framework would help evaluate the outcomes and impacts of tourism projects including distilling insights for possible replication.

APPENDIX

APPENDIX 1 APPROVED TOURISM-RELATED TECHNICAL ASSISTANCE AND PROJECTS

NAME OF TECHNICAL ASSISTANCE	/ AMOUNT (US\$ '000)
RETA 6279 GMS: Preparing the Sustainable Tourism Development Project	900
RETA 6225 South Asia Subregional Economic Cooperation Human Resource Development and Capacity Building	600
RETA 6179 Greater Mekong Subregion Tourism Sector Strategy	800
RETA 6131 South Asia Subregional Economic Cooperation Tourism Development Plan	450
RETA 5893 Mekong/Lancang River Tourism Infrastructure Development	600
RETA 5807 Tourism Skills Development in the GMS	125
RETA 5743 Mekong/Lancang River Tourism Planning Study	600
RETA 5647 Regional Program to Train Trainers in Tourism in the GMS	130
RETA 5440 Tourism Financing	100
TA 3454 Building Capacity in Tourism Planning	586
TA 3200 Strengthening Tourism Planning	150
TA 2685 Tourism Sector Development	600
TA 2483 Tourism Development	405
TA 2483 Tourism Development (Supplementary)	150
TA 2140 2nd Tourism Infrastructure Development	460
TA 1662 Tourism Development Study	295
TA 1432 National Tourism Plan	550
TA 1428 Tourism Development	100
TA 1384 Tourism Master Plan Study	275
TA 1298 Tourism Development	65
TA 1137 Tourism Development Program	460
TA 0991 Tourism Master Plan	375
NAME OF PROJECT	AMQUNT (US\$ million
NEP: Tourism Infrastructure Development (Loan 1156)	10.4
NEP: 2nd Tourism Development (Loan 1971)	17.2
GMS: Mekong Tourism Development Project (CAM/LAO/VIE) (Loans 1969/70/71)	35.0

REFERENCES

- ADB. 2005. Greater Mekong Subregion: Tourism Sector Strategy. Manila.
- Archer, B. H. 1995. "Importance Of Tourism For The Economy Of Bermuda." Annals of Tourism Research 22:918-30.
- Archer, B., and J. Fletcher. 1996. "The Economic Impact of Tourism in the Seychelles." Annals of Tourism Research 23(1):32-47.
- Armstrong, W. E., S. Daniel, and A. A. Franci. 1974. "A Structural Analysis of the Barbados Economy, 1968, with an Application to the Tourist Industry." Social and Economic Studies 23:493-520.
- Ashley, C., Roe, D. and H. Goodwin. 2001b. "Pro-poor Tourism Strategies: Making Tourism Work for the Poor." In Overseas Development Institute, Pro-Poor Tourism Report No. 1. Nottingham: Russell
- Bendavid-Val, A. 1983. Regional and Local Economic Analysis for Practitioners. New York: Praeger.
- Bergstrom, J. C., H. K. Cordell, G. A. Ashley, and A. E. Watson. 1990. "Economic Impacts of Recreational Spending on Rural Areas: A Case Study." Economic Development Quarterly 4(1):29-39.
- Blakely, E. J. 1994. Planning Local Economic Development: Theory and Practice, 2nd ed. Thousand Oaks, CA: SAGE Publications, Inc.
- Britton, S. 1981. "The Spatial Organisation of Tourism in a Neo-colonial Economy: A Fiji Case Study. Pacific Viewpoint 21(2):144-65.
- Brohman, J. 1996. "New Directions in Tourism for Third World Development." Annals of Tourism Research 23(1):48-78.
- Calantone, R., et.al. 1987. "A Comprehensive Review of the Tourism Forecasting Literature." Journal of Travel Research 26 (2): 23-39.
- Crick, M. 1989. "Representations of International Tourism in the Social Sciences: Sun, Sex, Sights, Savings, and Servility." Annual Review of Anthropology 18:307-44.
- Curry, S. 1986. "The Economic Impact of the Tourist Industry in the United Republic of Tanzania: An Input-Output Analysis." Industry and Development 19:55-75.
- Dawson, S., D. Blahna, and J. Keith. 1993. "Expected and Actual Regional Economic Impacts of Great Basin National Park." Journal of Park and Recreation Administration 11(1):45-57.
- Department for International Development (DFID). 1999a. Sustainable Tourism and Poverty Elimination Study. Report for the Department for International Development by Deloitte & Touche, IIED, and ODI, London.
- Diamond, J. 1976. "Tourism and Development Policy: A Quantitative Appraisal." Bulletin of Economic Research 28:36-80.
- Douglas, A. J., and D. A. Harpman. 1995. "Estimating Recreation Employment Effects with IMPLAN for the Glen Canyon Dam Region." Journal of Environmental Management 44:233-47
- Feder, G. 1983. "On Exports and Economic Growth." Journal of Development Economics 12:59-73
- Frechtling, D. C. 1994. "Assessing the Economic Impacts of Travel and Tourism Introduction to Travel Economic Impact Estimation." In J. R. Brent Ritchie and C. R. Goeldner, eds., Travel, Tourism and Hospitality Research: A Handbook for Managers and Researchers, 2nd ed. Singapore: John Wiley & Sons (Asia) Pte Ltd.
- Goodwin, H. 2000. "Pro-Poor Tourism, Opportunities for Sustainable Local Development." Development and Cooperation 5:12-4.
- Hanson, K., S. Robinson, and S. Tokarick. 1990. U.S. Adjustment in 1990's-A CGE Analysis of Alternative Trade Strategy. United States Department of Agriculture.
- Harrison, D., ed. 1992. Tourism and the Less Developed Countries. London: Wiley.
- Hemmati, M. (ed.) 1999. Gender and Tourism: Women's Employment and Participation. Report submitted to the UN CSD 7th Session. UNED, London.

- Heng, T. M., and L. Low. 1990. "Economic Impact of Tourism in Singapore." *Annals of Tourism Research* 17:246–69.
- Institute for Tourism and Recreation Research. 1996. "Nonresident Travel to Montana." Available: http://www.forestry.umt.edu/itrr/sub/IMPACT.HTM.
- Johnson, R. L., F. Obrermiller, and H. Radtke. 1989. "The Economic Impact of Tourism Sales." *Journal of Leisure Research* 21(2):140-54.
- Leistritz, F. L., and S. H. Murdock. 1981. The Socioeconomic Impact of Resource Development: Methods for Assessment. Boulder, CO: Westview Press.
- Leitch, J. A., and F. L. Leistritz. 1985. "Techniques for Assessing the Secondary Impact of Recreation and Tourism." In D.B. Propst, Compiler, Assessing the Economic Impacts of Recreation and Tourism.

 Southeastern Forest Experiment Station, North Carolina.
- Linstone H. A., and M. Turoff.1975. The Delphi Method: Techniques and Applications. Addison-Wesley.
- Liu, J. 1986. "Relative Economic Contribution of Visitor Groups in Hawaii." *Journal of Travel Research* 15:2–13.
- Liu, J., and T. Var. 1982. "Differential Multipliers for the Accommodation Sector." *Tourism Management* 3:117–87.
- Longwoods International. 1992. The Economic Impact, Performance and Profile of The New Jersey Travel and Tourism Industry 1990-1991. New Jersey Department of Commerce and Economic Department.
- ______. 1993. The Economic Impact, Performance and Profile of The New Jersey Travel and Tourism Industry 1991-1992. New Jersey Department of Commerce and Economic Department.
- ______. 1995. The Economic Impact of The Connecticut Travel and Tourism Industry 1992–1993. Connecticut Department of Economic Development.
- Miller, R. E., and P. D. Blair. 1985. *Input-Output Analysis: Foundations and Extensions*. New Jersey: Prentice-Hall.
- Modeste, N. C. 1995. "The Impact of Growth in the Tourism Section On Economic Development: The Experience of Selected Caribbean Countries." *Economia Internazionale* 48(3, August):375–85.
- Mowforth, M., and I. Munt. 1998. *Tourism and Sustainability: New Tourism in the Third World*. New York: Routledge.
- Oppenheim, N. 1980. Applied Models in Urban and Regional Analysis. Englewood Cliffs, NJ: Prentice-Hall, Inc.
- Pabuayon, I. 1991. Methodologies for Philippine Wildlife and Protected Areas. Technical Report No.10, Natural Resources Accounting Project, Department of Environment and Natural Resources, Manila.
- Petropoulos, et. al. 2005. "A Technical Analysis Approach to Tourism Demand Forecasting." Applied Economic Letters 12(6):327-33.
- Pfister, R. L. 1976. "On Improving Export Base Studies." Regional Science Perspectives 6:104-13.
- Pleeter, S. 1980. Economic Impact Analysis: Methodology and Applications. Boston: Martinus Nijhoff Publishing.
- Ram, R. 1986. "Government Size and Economic Growth: A New Framework and Some Evidence from Cross-Section and Time-Series." *American Economic Review* 76:191–203.
- Roe, D., and P. Urquhart Khanya. 2001. "Pro-poor Tourism: Harnessing the World's Largest Industry for the World's Poor." Paper prepared for the World Summit on Sustainable Development, Johannesburg. International Institute for Environment and Development and Regional and International Networking Group, London.
- Roe, D., J. Goodwin, and C. Ashley. 2002. The Tourism Industry and Poverty Reduction: A Business Primer. Pro-poor Tourism Briefing No. 2, Centre for Responsible Tourism, International Institute for Environment, and Development and Overseas Development Institute, London.
- Rowe, G. and G. Wright. 1999. "The Delphi Technique as a Forecasting Tool: Issues and Analysis." *International Journal of Forecasting* 15(4):353-75.

TOURISM FOR PRO-POOR AND SUSTAINABLE GROWTH: ECONOMIC ANALYSIS OF TOURISM PROJECTS TUN LIN AND FRANKLIN D. DE GUZMAN

- Seow, G. 1981. "Economic Impact Significance of Tourism in Singapore." Singapore Economic Review 26:64-79.
- Shah, K. 2000. "Tourism, the Poor and Other Stakeholders: Asian Experience." ODI Fair-Trade in Tourism Paper, Overseas Development Institute, London.
- Stokey, E., and R. J. Zeckhauser. 1978. A Primer for Policy Analyses. New York: Norton.
- Sugden, R., and A. Williams. 1978. The Principles of Practical Cost-Benefit Analysis. Oxford: Oxford University Press.
- Tanjuakio, R. V., S. E. Hastings, and P. J. Tytus. 1996. "The Economic Contribution of Agriculture in Delaware." Agricultural and Resource Economics Review 25(1):46-53.
- Thorbecke, E. 1985. "The Social Accounting Matrix and Consistent-Type Planning Models." In G. Pyatt and J. I. Round, eds., Social Accounting Matrices: A Basis for Planning. World Bank, Washington, DC.
- Tooman, L. A. 1997. "Multipliers and Life Cycles: A Comparison of Methods for Evaluating Tourism and Its Impacts." *Journal of Economic Issues* 31(4, December):917–32.
- Wagner, J. E. 1997. "Estimating the Economic Impacts of Tourism." Annals of Tourism Research 24(3):592-608.
- Weaver T. 1972. Delphi: A Critical Review. School of Education Policy Research Center, Syracuse University, New York.
- Witt, S. F., and C. A. Witt. 1995. "Forecasting Tourism Demand: A Review of Empirical Research." *International Journal of Forecasting* 11(3):447-75.
- Woudenberg F. 1991. "An Evaluation of Delphi." Technological Forecasting and Social Change 40:131-50
- WTO/OMT. 2002. Tourism and Poverty Alleviation Report. World Tourism Organisation, Madrid.
- WTO and UNCTAD. 2001. Tourism in the Least Developed Countries. World Tourism Organization, Madrid.
- Zhou, D., J. F. Yanagida, U. Chakravorty, and P. S. Leung. 1997. "Estimating Economic Impacts from Tourism." *Annals of Tourism Research* 24(1):76–89.
- Zinder, H. 1969. The Future of Tourism in the Eastern Caribbean. Zinder and Associates, Washington DC.

PUBLICATIONS FROM THE ECONOMICS AND RESEARCH DEPARTMENT

ERD TECHNICAL NOTE SERIES (TNS)

(Published in-house; Available through ADB Office of External Relations; Free of Charge)

No. 1	Contingency Calculations for Environmental Impacts with Unknown Monetary Values —David Dole, February 2002	No. 12	Improving the Relevance and Feasibility of Agriculture and Rural Development Operational Designs: How Economic Analyses Can Help
No. 2	Integrating Risk into ADB's Economic Analysis		-Richard Bolt, September 2005
	of Projects	No. 13	Assessing the Use of Project Distribution and
	—Nigel Rayner, Anneli Lagman-Martin, and Keith Ward, June 2002		Poverty Impact Analyses at the Asian Developmen
No. 3	Measuring Willingness to Pay for Electricity		-Franklin D. De Guzman, October 2005
	-Peter Choynowski, July 2002	No. 14	Assessing Aid for a Sector Development Plan:
No. 4	Economic Issues in the Design and Analysis of a		Economic Analysis of a Sector Loan
	Wastewater Treatment Project		—David Dole, November 2005
	-David Dole, July 2002	No. 15	Debt Management Analysis of Nepal's Public Deb
No. 5	An Analysis and Case Study of the Role of	110. 10	-Sungsup Ra, Changyong Rhee, and Joon-Ho
110. 0	Environmental Economics at the Asian		Hahm, December 2005
	Development Bank	No. 16	Evaluating Microfinance Program Innovation with
	—David Dole and Piya Abeygunawardena,	110. 10	Randomized Control Trials: An Example from
	September 2002		Group Versus Individual Lending
No. 6	Economic Analysis of Health Projects: A Case Study		-Xavier Giné, Tomoko Harigaya, Dean Karlan, an
140. 0	in Cambodia		Binh T. Nguyen, March 2006
	—Erik Bloom and Peter Choynowski, May 2003	No. 17	Setting User Charges for Urban Water Supply: A
No. 7	Strengthening the Economic Analysis of Natural		Case Study of the Metropolitan Cebu Water
110. 1	Resource Management Projects		District in the Philippines
	-Keith Ward, September 2003		—David Dole and Edna Balucan, June 2006
No. 8	Testing Savings Product Innovations Using an	No. 18	Forecasting Inflation and GDP Growth: Automatic
140. 0	Experimental Methodology	140. 10	Leading Indicator (ALI) Method versus Macro
	—Nava Ashraf, Dean S. Karlan, and Wesley Yin,		Econometric Structural Models (MESMs)
	November 2003		-Marie Anne Cagas, Geoffrey Ducanes, Nedelyn
No. 9	Setting User Charges for Public Services: Policies		Magtibay-Ramos, Duo Qin and Pilipinas Quising,
140. 5	and Practice at the Asian Development Bank		July 2006
	-David Dole, December 2003	No. 19	Willingness-to-Pay and Design of Water Supply
No. 10	Beyond Cost Recovery: Setting User Charges for	140. 15	and Sanitation Projects: A Good Practice Case
140. 10	Financial, Economic, and Social Goals		Study
	—David Dole and Ian Bartlett, January 2004		-Herath Gunatilake, Jui-Chen Yang, Subhrendu
No. 11	Shadow Exchange Rates for Project Economic		Pattanayak, and Caroline van den Berg, December
140. 11	Analysis: Toward Improving Practice at the Asian		2006
	Development Bank	No. 20	Tourism for Pro-Poor and Sutainable Growth:
	—Anneli Lagman-Martin, February 2004	110. 20	Economic Analysis of Tourism Projects
	-Timen Laginarianan, Periaday 2003		—Tun Lin and Franklin D. De Guzman,
			January 2007
			Samuely 2001

ERD POLICY BRIEF SERIES (PBS)

(Published in-house; Available through ADB Office of External Relations; Free of charge)

iciency and in Cambodia
Brad Schwartz,
Crisis: What
el Tell Us?
wling,
angements
erre Verbiest,
rspective
and Lea
le T

No. 10	Is Economic Openness Good for Regional	No. 30	Living with Higher Interest Rates: Is Asia Ready?
	Development and Poverty Reduction? The		-Cyn-Young Park, August 2004
	Philippines	No. 31	Reserve Accumulation, Sterilization, and Policy
	—E. M. Pernia and Pilipinas Quising, October		Dilemma —Akiko Terada-Hagiwara, October 2004
M- 11	2002	No. 32	The Primacy of Reforms in the Emergence of
No. 11	Implications of a US Dollar Depreciation for Asian	140. 32	People's Republic of China and India
	Developing Countries		
	—Emma Fan, July 2002		—Ifzal Ali and Emma Xiaoqin Fan, November
No. 12	Dangers of Deflation		2004
	—D. Brooks and Pilipinas Quising, December 2002	No. 33	Population Health and Foreign Direct Investment:
No. 13	Infrastructure and Poverty Reduction—		Does Poor Health Signal Poor Government
	What is the Connection?		Effectiveness?
	—Ifzal Ali and Ernesto Pernia, January 2003		—Ajay Tandon, January 2005
No. 14	Infrastructure and Poverty Reduction—	No. 34	Financing Infrastructure Development: Asian
	Making Markets Work for the Poor		Developing Countries Need to Tap Bond Markets
	-Xianbin Yao, May 2003		More Rigorously
No. 15	SARS: Economic Impacts and Implications		-Yun-Hwan Kim, February 2005
	-Emma Xiaogin Fan, May 2003	No. 35	Attaining Millennium Development Goals in
No. 16	Emerging Tax Issues: Implications of Globalization		Health: Isn't Economic Growth Enough?
20	and Technology		-Ajay Tandon, March 2005
	-Kanokpan Lao Araya, May 2003	No. 36	Instilling Credit Culture in State-owned Banks-
No. 17	Pro-Poor Growth: What is It and Why is It	110. 00	Experience from Lao PDR
110. 11	Important?		-Robert Boumphrey, Paul Dickie, and Samiuela
	—Ernesto M. Pernia, May 2003		Tukuafu, April 2005
N- 10		No. 37	Coping with Global Imbalances and Asian
No. 18	Public-Private Partnership for Competitiveness	140. 51	Currencies
	—Jesus Felipe, June 2003		
No. 19	Reviving Asian Economic Growth Requires Further		-Cyn-Young Park, May 2005
	Reforms	No. 38	Asia's Long-term Growth and Integration:
	—Ifzal Ali, June 2003		Reaching beyond Trade Policy Barriers
No. 20	The Millennium Development Goals and Poverty:		-Douglas H. Brooks, David Roland-Holst, and Fan
	Are We Counting the World's Poor Right?		Zhai, September 2005
	-M. G. Quibria, July 2003	No. 39	Competition Policy and Development
No. 21	Trade and Poverty: What are the Connections?		-Douglas H. Brooks, October 2005
	—Douglas H. Brooks, July 2003	No. 40	Highlighting Poverty as Vulnerability: The 2005
No. 22	Adapting Education to the Global Economy		Earthquake in Pakistan
	—Olivier Dupriez, September 2003		-Rana Hasan and Ajay Tandon, October 2005
No. 23	Avian Flu: An Economic Assessment for Selected	No. 41	Conceptualizing and Measuring Poverty as
	Developing Countries in Asia		Vulnerability: Does It Make a Difference?
	—Jean-Pierre Verbiest and Charissa Castillo,		-Ajay Tandon and Rana Hasan, October 2005
	March 2004	No. 42	Potential Economic Impact of an Avian Flu
No. 25	Purchasing Power Parities and the International		Pandemic on Asia
	Comparison Program in a Globalized World		-Erik Bloom, Vincent de Wit, and Mary Jane
	-Bishnu Pant, March 2004		Carangal-San Jose, November 2005
No. 26	A Note on Dual/Multiple Exchange Rates	No. 43	Creating Better and More Jobs in Indonesia: A
140. 20	—Emma Xiaoqin Fan, May 2004	110. 40	Blueprint for Policy Action
No. 27	Inclusive Growth for Sustainable Poverty Reduction		-Guntur Sugiyarto, December 2005
140. 27		No. 44	The Challenge of Job Creation in Asia
	in Developing Asia: The Enabling Role of	140. 44	
	Infrastructure Development	NT- 45	—Jesus Felipe and Rana Hasan, April 2006
	—Ifzal Ali and Xianbin Yao, May 2004	No. 45	International Payments Imbalances
No. 28	Higher Oil Prices: Asian Perspectives and		—Jesus Felipe, Frank Harrigan, and Aashish
	Implications for 2004-2005		Mehta, April 2006
	-Cyn-Young Park, June 2004	No. 46	Improving Primary Enrollment Rates among
No. 29	Accelerating Agriculture and Rural Development for		the Poor
	Inclusive Growth: Policy Implications for		—Ajay Tandon, August 2006
	Developing Asia		
	-Richard Bolt, July 2004		

the Uruguay Round

ERD WORKING PAPER SERIES (WPS)(Published in-house; Available through ADB Office of External Relations; Free of Charge)

No. 1	Capitalizing on Globalization		-J. Michael Finger, September 2002
	-Barry Eichengreen, January 2002	No. 22	Conceptual Issues in the Role of Education
No. 2	Policy-based Lending and Poverty Reduction:		Decentralization in Promoting Effective Schooling in Asian Developing Countries
	An Overview of Processes, Assessment and Options		—Jere R. Behrman, Anil B. Deolalikar, and Lee-
	—Richard Bolt and Manabu Fujimura, January		Ying Son, September 2002
	2002	No. 23	Promoting Effective Schooling through Education
No. 3	The Automotive Supply Chain: Global Trends		Decentralization in Bangladesh, Indonesia, and
	and Asian Perspectives		Philippines
	Francisco Veloso and Rajiv Kumar, January 2002		—Jere R. Behrman, Anil B. Deolalikar, and Lee-
No. 4	International Competitiveness of Asian Firms:	No. 24	Ying Son, September 2002 Financial Opening under the WTO Agreement in
	An Analytical Framework —Rajiv Kumar and Doren Chadee, February 2002	140. 24	Selected Asian Countries: Progress and Issues
No. 5	The International Competitiveness of Asian		-Yun-Hwan Kim, September 2002
110. 0	Economies in the Apparel Commodity Chain	No. 25	Revisiting Growth and Poverty Reduction in
	—Gary Gereffi, February 2002		Indonesia: What Do Subnational Data Show?
No. 6	Monetary and Financial Cooperation in East		—Arsenio M. Balisacan, Ernesto M. Pernia,
	Asia—The Chiang Mai Initiative and Beyond	No. 26	and Abuzar Asra, October 2002 Causes of the 1997 Asian Financial Crisis: What
No. 7	—Pradumna B. Rana, February 2002 Probing Beneath Cross-national Averages: Poverty,	140. 20	Can an Early Warning System Model Tell Us?
No. 7	Inequality, and Growth in the Philippines		-Juzhong Zhuang and J. Malcolm Dowling,
	-Arsenio M. Balisacan and Ernesto M. Pernia,		October 2002
	March 2002	No. 27	Digital Divide: Determinants and Policies with
No. 8	Poverty, Growth, and Inequality in Thailand		Special Reference to Asia
	-Anil B. Deolalikar, April 2002		-M. G. Quibria, Shamsun N. Ahmed, Ted Tschang, and Mari-Len Reyes-Macasaquit, October
No. 9	Microfinance in Northeast Thailand: Who Benefits		2002
	and How Much? —Brett E. Coleman, April 2002	No. 28	Regional Cooperation in Asia: Long-term Progress,
No. 10	Poverty Reduction and the Role of Institutions in	20	Recent Retrogression, and the Way Forward
1.0. 10	Developing Asia		—Ramgopal Agarwala and Brahm Prakash,
	-Anil B. Deolalikar, Alex B. Brilliantes, Jr.,		October 2002
	Raghav Gaiha, Ernesto M. Pernia, Mary Racelis	No. 29	How can Cambodia, Lao PDR, Myanmar, and Vie
	with the assistance of Marita Concepcion Castro-		Nam Cope with Revenue Lost Due to AFTA Tariff
	Guevara, Liza L. Lim, Pilipinas F. Quising, May		Reductions? —Kanokpan Lao-Araya, November 2002
No. 11	2002 The European Social Model: Lessons for	No. 30	Asian Regionalism and Its Effects on Trade in the
140. 11	Developing Countries	2.0. 00	1980s and 1990s
	-Assar Lindbeck, May 2002		-Ramon Clarete, Christopher Edmonds, and
No. 12	Costs and Benefits of a Common Currency for		Jessica Seddon Wallack, November 2002
	ASEAN	No. 31	New Economy and the Effects of Industrial
	—Srinivasa Madhur, May 2002		Structures on International Equity Market Correlations
No. 13			—Cyn-Young Park and Jaejoon Woo, December
No. 14	—Raul Fabella, May 2002 Toward A Political Economy Approach		2002
140. 14	to Policy-based Lending	No. 32	Leading Indicators of Business Cycles in Malaysia
	-George Abonyi, May 2002		and the Philippines
No. 15	A Framework for Establishing Priorities in a		—Wenda Zhang and Juzhong Zhuang, December
	Country Poverty Reduction Strategy		2002
	—Ron Duncan and Steve Pollard, June 2002	No. 33	Technological Spillovers from Foreign Direct
No. 16	The Role of Infrastructure in Land-use Dynamics		Investment—A Survey —Emma Xiaoqin Fan, December 2002
	and Rice Production in Viet Nam's Mekong River Delta	No. 34	Economic Openness and Regional Development in
	-Christopher Edmonds, July 2002		the Philippines
No. 17	Effect of Decentralization Strategy on		—Ernesto M. Pernia and Pilipinas F. Quising,
	Macroeconomic Stability in Thailand		January 2003
	—Kanokpan Lao-Araya, August 2002	No. 35	Bond Market Development in East Asia:
No. 18	Poverty and Patterns of Growth		Issues and Challenges
	—Rana Hasan and M. G. Quibria, August 2002		—Raul Fabella and Srinivasa Madhur, January 2003
No. 19	Why are Some Countries Richer than Others? A Reassessment of Mankiw-Romer-Weil's Test of	No. 36	Environment Statistics in Central Asia: Progress
	the Neoclassical Growth Model	110. 00	and Prospects
	—Jesus Felipe and John McCombie, August 2002		-Robert Ballance and Bishnu D. Pant, March
No. 20	Modernization and Son Preserence in People's		2003
- •	Republic of China	No. 37	Electricity Demand in the People's Republic of
	-Robin Burgess and Juzhong Zhuang, September		China: Investment Requirement and
	2002		Environmental Impact
No. 21	The Doha Agenda and Development: A View from	M. 00	-Bo Q. Lin, March 2003 Foreign Direct Investment in Developing Asia:
	the Uruguay Round	No. 38	Loteign Ditect thresoment in Desciobing Usia:

	Trends, Effects, and Likely Issues for the Forthcoming WTO Negotiations —Douglas H. Brooks, Emma Xiaoqin Fan,
	and Lea R. Sumulong, April 2003
No. 39	The Political Economy of Good Governance for
140. 55	
	Poverty Alleviation Policies
	-Narayan Lakshman, April 2003
No. 40	The Puzzle of Social Capital
	A Critical Review
	-M. G. Quibria, May 2003
No. 41	Industrial Structure, Technical Change, and the
110. 11	Role of Government in Development of the
	Electronics and Information Industry in
	Taipei,China
	—Yeo Lin, May 2003
No. 42	Economic Growth and Poverty Reduction
	in Viet Nam
	-Arsenio M. Balisacan, Ernesto M. Pernia, and
	Gemma Esther B. Estrada, June 2003
No. 43	Why Has Income Inequality in Thailand
	Increased? An Analysis Using 1975-1998 Surveys
	—Taizo Motonishi, June 2003
No. 44	Welfare Impacts of Electricity Generation Sector
• • • • • • • • • • • • • • • • • • • •	Reform in the Philippines
	-Natsuko Toba, June 2003
No. 45	A Review of Commitment Savings Products in
	Developing Countries
	—Nava Ashraf, Nathalie Gons, Dean S. Karlan,
	and Wesley Yin, July 2003
No. 46	Local Government Finance, Private Resources,
	and Local Credit Markets in Asia
	-Roberto de Vera and Yun-Hwan Kim, October
	2003
No. 47	Excess Investment and Efficiency Loss During
	Reforms: The Case of Provincial-level Fixed-Asset
	Investment in People's Republic of China
	-Duo Qin and Haiyan Song, October 2003
No. 48	Is Export-led Growth Passe? Implications for
110. 40	
	Developing Asia
	—Jesus Felipe, December 2003
No. 49	Changing Bank Lending Behavior and Corporate
	Financing in Asia—Some Research Issues
	-Emma Xiaoqin Fan and Akiko Terada-Hagiwara,
	December 2003
No. 50	Is People's Republic of China's Rising Services
110. 00	Sector Leading to Cost Disease?
	.
	—Duo Qin, March 2004
No. 51	Poverty Estimates in India: Some Key Issues
	—Savita Sharma, May 2004
No. 52	Restructuring and Regulatory Reform in the Power
	Sector: Review of Experience and Issues
	-Peter Choynowski, May 2004
NT- E0	Competitiveness, Income Distribution, and Growth
No. 53	Competitiveness, income Distribution, and Growni
	in the Philippines: What Does the Long-run
	Evidence Show?
	—Jesus Felipe and Grace C. Sipin, June 2004
No. 54	Practices of Poverty Measurement and Poverty
	Profile of Bangladesh
	-Faizuddin Ahmed, August 2004
No. 55	Experience of Asian Asset Management
140. 55	
	Companies: Do They Increase Moral Hazard?
	-Evidence from Thailand
	—Akiko Terada-Hagiwara and Gloria Pasadilla,
	September 2004
No. 56	Viet Nam: Foreign Direct Investment and
	Postcrisis Regional Integration
	-Vittorio Leproux and Douglas H. Brooks,
	September 2004
No. 57	Practices of Poverty Measurement and Poverty
	Profile of Nepal
	Devendra Chhetry, September 2004
No. 58	Monetary Poverty Estimates in Sri Lanka:
	Selected Issues
	—Neranjana Gunetilleke and Dinushka
	Senanayake, October 2004

No. 59	Labor Market Distortions, Rural-Urban Inequality, and the Opening of People's Republic of China's Economy
	-Thomas Hertel and Fan Zhai, November 2004
No. 60	Measuring Competitiveness in the World's Smallest
	Economies: Introducing the SSMECI
	-Ganeshan Wignaraja and David Joiner, November
N. 01	2004
No. 61	Foreign Exchange Reserves, Exchange Rate Regimes, and Monetary Policy: Issues in Asia
	—Akiko Terada-Hagiwara, January 2005
No. 62	A Small Macroeconometric Model of the Philippine
110. 02	Economy
	-Geoffrey Ducanes, Marie Anne Cagas, Duo Qin,
	Pilipinas Quising, and Nedelyn Magtibay-Ramos,
	January 2005
No. 63	Developing the Market for Local Currency Bonds
	by Foreign Issuers: Lessons from Asia
No. 64	—Tobias Hoschka, February 2005 Empirical Assessment of Sustainability and
NO. 04	Feasibility of Government Debt: The Philippines
	Case
	—Duo Qin, Marie Anne Cagas, Geoffrey Ducanes,
	Nedelyn Magtibay-Ramos, and Pilipinas Quising,
•	February 2005
No. 65	Poverty and Foreign Aid
	Evidence from Cross-Country Data
	—Abuzar Asra, Gemma Estrada, Yangseom Kim, and M. G. Quibria, March 2005
No. 66	Measuring Efficiency of Macro Systems: An
140. 00	Application to Millennium Development Goal
	Attainment
	-Ajay Tandon, March 2005
No. 67	Banks and Corporate Debt Market Development
	—Paul Dickie and Emma Xiaoqin Fan, April 2005
No. 68	Local Currency Financing—The Next Frontier for MDBs?
	—Tobias C. Hoschka, April 2005
No. 69	Export or Domestic-Led Growth in Asia?
	—Jesus Felipe and Joseph Lim, May 2005
No. 70	Policy Reform in Viet Nam and the Asian
	Development Bank's State-owned Enterprise Reform and Corporate Governance Program Loan
	-George Abonyi, August 2005
No. 71	Policy Reform in Thailand and the Asian Develop-
	ment Bank's Agricultural Sector Program Loan
	-George Abonyi, September 2005
No. 72	Can the Poor Benefit from the Doha Agenda? The
	Case of Indonesia —Douglas H. Brooks and Guntur Sugiyarto,
	October 2005
No. 73	Impacts of the Doha Development Agenda on
	People's Republic of China: The Role of
	Complementary Education Reforms
	-Fan Zhai and Thomas Hertel, October 2005
No. 74	Growth and Trade Horizons for Asia: Long-term
	Forecasts for Regional Integration
	—David Roland-Holst, Jean-Pierre Verbiest, and Fan Zhai, November 2005
No. 75	Macroeconomic Impact of HIV/AIDS in the Asian
2.00	and Pacific Region
	-Ajay Tandon, November 2005
No. 76	Policy Reform in Indonesia and the Asian
	Development Bank's Financial Sector Governance
	Reforms Program Loan
No. 77	—George Abonyi, December 2005 Dynamics of Manufacturing Competitiveness in
110. 21	South Asia: ANalysis through Export Data
	-Hans-Peter Brunner and Massimiliano Cali,
	December 2005
No. 78	Trade Facilitation
	-Teruo Ujiie, January 2006
No. 79	An Assessment of Cross-country Fiscal
	Consolidation —Bruno Carrasco and Seung Mo Choi,

A.LT. LIBRAR

February 2006

No. 80 Central Asia: Mapping Future Prospects to 2015

—Malcolm Dowling and Ganeshan Wignaraja,
April 2006

No. 81 A Small Macroeconometric Model of the People's Republic of China

—Duo Qin, Marie Anne Cagas, Geoffrey Ducanes, Nedelyn Magtibay-Ramos, Pilipinas Quising, Xin-Hua He, Rui Liu, and Shi-Guo Liu, June 2006

No. 82 Institutions and Policies for Growth and Poverty Reduction: The Role of Private Sector Development —Rana Hasan, Devashish Mitra, and Mehmet Ulubasoglu. July 2006

Ulubasoglu, July 2006

No. 83 Preferential Trade Agreements in Asia:
Alternative Scenarios of "Hub and Spoke"
—Fan Zhai, October 2006

No. 84 Income Disparity and Economic Growth: Evidence from People's Republic of China

— Duo Qin, Marie Anne Cagas, Geoffrey Ducanes, Xinhua He, Rui Liu, and Shiguo Liu, October 2006

No. 85 Macroeconomic Effects of Fiscal Policies: Empirical Evidence from Bangladesh, People's Republic of China, Indonesia, and Philippines

— Geoffrey Ducanes, Marie Anne Cagas, Duo Qin,
Pilipinas Quising, and Mohammad Abdur
Razzaque, November 2006

No. 86 Economic Growth, Technological Change, and Patterns of Food and Agricultural Trade in Asia

— Thomas W. Hertel, Carlos E. Ludena, and Alla Golub, November 2006

No. 87 Expanding Access to Basic Services in Asia and the Pacific Region: Public-Private Partnerships for Poverty Reduction

— Adrian T. P. Panggabean, November 2006

No. 88 Income Volatility and Social Protection in Developing Asia

—Vandana Sipahimalani-Rao, November 2006

No. 89 Rules of Origin: Conceptual Explorations and Lessons from the Generalized System of Preferences

—Teruo Ujiie, December 2006

No. 90 Asia's Imprint on Global Commodity Markets

—Cyn-Young Park and Fan Zhai, December 2006

SPECIAL STUDIES, COMPLIMENTARY (Available through ADB Office of External Relations)

- Improving Domestic Resource Mobilization Through Financial Development: Overview September 1985
- Improving Domestic Resource Mobilization Through Financial Development: Bangladesh July 1986
- Improving Domestic Resource Mobilization Through Financial Development: Sri Lanka April 1987
- Improving Domestic Resource Mobilization Through Financial Development: India December 1987
- Financing Public Sector Development Expenditure in Selected Countries: Overview January 1988
- 6. Study of Selected Industries: A Brief Report
 April 1988
- Financing Public Sector Development Expenditure in Selected Countries: Bangladesh June 1988
- Financing Public Sector Development Expenditure in Selected Countries: India June 1988
- Financing Public Sector Development Expenditure in Selected Countries: Indonesia June 1988
- Financing Public Sector Development Expenditure in Selected Countries: Nepal June 1988
- Financing Public Sector Development Expenditure in Selected Countries: Pakistan June 1988
- Financing Public Sector Development Expenditure in Selected Countries: Philippines June 1988
- 13. Financing Public Sector Development Expenditure in Selected Countries: Thailand June 1988
- Towards Regional Cooperation in South Asia:
 ADB/EWC Symposium on Regional Cooperation in South Asia February 1988
- 15. Evaluating Rice Market Intervention Policies: Some Asian Examples April 1988
- Improving Domestic Resource Mobilization Through Financial Development: Nepal November 1988
- 17. Foreign Trade Barriers and Export Growth September 1988

- The Role of Small and Medium-Scale Industries in the Industrial Development of the Philippines April 1989
- The Role of Small and Medium-Scale Manufacturing Industries in Industrial Development: The Experience of Selected Asian Countries January 1990
- 20. National Accounts of Vanuatu, 1983-1987 January 1990
- National Accounts of Western Samoa, 1984-1986 February 1990
- 22. Human Resource Policy and Economic Development: Selected Country Studies July 1990
- 23. Export Finance: Some Asian Examples September 1990
- National Accounts of the Cook Islands, 1982-1986
 September 1990
- Framework for the Economic and Financial Appraisal of Urban Development Sector Projects January 1994
- Framework and Criteria for the Appraisal and Socioeconomic Justification of Education Projects January 1994
- 27. Investing in Asia 1997 (Co-published with OECD)
- The Future of Asia in the World Economy 1998 (Copublished with OECD)
- Financial Liberalisation in Asia: Analysis and Prospects 1999 (Co-published with OECD)
- Sustainable Recovery in Asia: Mobilizing Resources for Development 2000 (Co-published with OECD)
- Technology and Poverty Reduction in Asia and the Pacific 2001 (Co-published with OECD)
- 32. Asia and Europe 2002 (Co-published with OECD)
- 33. Economic Analysis: Retrospective 2003
- 34. Economic Analysis: Retrospective: 2003 Update 2004
- Development Indicators Reference Manual: Concepts and Definitions 2004

OLD MONOGRAPH SERIES

(Available through ADB Office of External Relations; Free of charge)

EDRC REPORT SERIES (ER)

No. 1	ASEAN and the Asian Development Bank —Seiji Naya, April 1982		Changes in Nine Asian LDCs —Pradumna B. Rana and J. Malcolm Dowling, Jr.,
No. 2	Development Issues for the Developing East		December 1983
	and Southeast Asian Countries	No. 22	Effects of External Shocks on the Balance
	and International Cooperation		of Payments, Policy Responses, and Debt
	—Seiji Naya and Graham Abbott, April 1982		Problems of Asian Developing Countries
No. 3	Aid, Savings, and Growth in the Asian Region		—Seiji Naya, December 1983
	-J. Malcolm Dowling and Ulrich Hiemenz,	No. 23	Changing Trade Patterns and Policy Issues:
	April 1982		The Prospects for East and Southeast Asian
No. 4	Development-oriented Foreign Investment		Developing Countries
	and the Role of ADB		-Seiji Naya and Ulrich Hiemenz, February 1984
	—Kiyoshi Kojima, April 1982	No. 24	Small-Scale Industries in Asian Economic
No. 5	The Multilateral Development Banks		Development: Problems and Prospects
	and the International Economy's Missing		—Seiji Naya, February 1984
	Public Sector	No. 25	A Study on the External Debt Indicators
	-John Lewis, June 1982		Applying Logit Analysis
No. 6	Notes on External Debt of DMCs		-Jungsoo Lee and Clarita Barretto, February 1984
	-Evelyn Go, July 1982	No. 26	Alternatives to Institutional Credit Programs
No. 7	Grant Element in Bank Loans		in the Agricultural Sector of Low-Income
	—Dal Hyun Kim, July 1982		Countries
No. 8	Shadow Exchange Rates and Standard		—Jennifer Sour, March 1984
	Conversion Factors in Project Evaluation	No. 27	Economic Scene in Asia and Its Special Features
	-Peter Warr, September 1982		—Kedar N. Kohli, November 1984
No. 9	Small and Medium-Scale Manufacturing	No. 28	The Effect of Terms of Trade Changes on the
	Establishments in ASEAN Countries:		Balance of Payments and Real National
	Perspectives and Policy Issues		Income of Asian Developing Countries
	—Mathias Bruch and Ulrich Hiemenz, January		-Jungsoo Lee and Lutgarda Labios, January 1985
	1983	No. 29	Cause and Effect in the World Sugar Market:
No. 10	A Note on the Third Ministerial Meeting of GATT		Some Empirical Findings 1951-1982
	-Jungsoo Lee, January 1983		—Yoshihiro Iwasaki, February 1985
No. 11	Macroeconomic Forecasts for the Republic	No. 30	Sources of Balance of Payments Problem
	of China, Hong Kong, and Republic of Korea		in the 1970s: The Asian Experience
	—J.M. Dowling, January 1983		—Pradumna Rana, February 1985
No. 12		No. 31	India's Manufactured Exports: An Analysis
	—Seiji Naya, March 1983		of Supply Sectors
No. 13	The Future Prospects for the Developing		—Ifzal Ali, February 1985
	Countries of Asia	No. 32	Meeting Basic Human Needs in Asian
	—Seiji Naya, March 1983		Developing Countries
No. 14	Energy and Structural Change in the Asia-		-Jungsoo Lee and Emma Banaria, March 1985
	Pacific Region, Summary of the Thirteenth	No. 33	The Impact of Foreign Capital Inflow
	Pacific Trade and Development Conference		on Investment and Economic Growth
	—Seiji Naya, March 1983		in Developing Asia
No. 15	A Survey of Empirical Studies on Demand		-Evelyn Go, May 1985
	for Electricity with Special Emphasis on Price	No. 34	
	Elasticity of Demand		in the Pacific and Asian Region:
	—Wisarn Pupphavesa, June 1983		Priorities and Perspectives
No. 16	Determinants of Paddy Production in Indonesia:		-V.V. Desai, April 1986
	1972-1981-A Simultaneous Equation Model	No. 35	Impact of Appreciation of the Yen on
	Approach		Developing Member Countries of the Bank
	—T.K. Jayaraman, June 1983		-Jungsoo Lee, Pradumna Rana, and Ifzal Ali,
No. 17	The Philippine Economy: Economic		May 1986
	Forecasts for 1983 and 1984	No. 36	Smuggling and Domestic Economic Policies
	—J.M. Dowling, E. Go, and C.N. Castillo, June		in Developing Countries
	1983		-A.H.M.N. Chowdhury, October 1986
No. 18	B Economic Forecast for Indonesia	No. 37	Public Investment Criteria: Economic Internal
	—J.M. Dowling, H.Y. Kim, Y.K. Wang,		Rate of Return and Equalizing Discount Rate
	and C.N. Castillo, June 1983		—Ifzal Ali, November 1986
No. 19	Relative External Debt Situation of Asian	No. 38	Review of the Theory of Neoclassical Political
	Developing Countries: An Application		Economy: An Application to Trade Policies
	of Ranking Method		-M.G. Quibria, December 1986
	-Jungsoo Lee, June 1983	No. 39	Factors Influencing the Choice of Location:
No. 20	New Evidence on Yields, Fertilizer Application,		Local and Foreign Firms in the Philippines
	and Prices in Asian Rice Production		-E.M. Pernia and A.N. Herrin, February 1987
	—William James and Teresita Ramirez, July 1983	No. 40	A Demographic Perspective on Developing
No 2	1 Inflationary Effects of Exchange Rate		Asia and Its Relevance to the Bank

	-E.M. Pernia, May 1987	No. 55	Financial Sector and Economic
No. 41	Emerging Issues in Asia and Social Cost		Development: A Survey
	Benefit Analysis		-Jungsoo Lee, September 1991
	—I. Ali, September 1988	No. 56	A Framework for Justifying Bank-Assisted
No. 42	Shifting Revealed Comparative Advantage:		Education Projects in Asia: A Review
	Experiences of Asian and Pacific Developing		of the Socioeconomic Analysis
	Countries		and Identification of Areas of Improvement
	—P.B. Rana, November 1988		-Etienne Van De Walle, February 1992
No. 43	Agricultural Price Policy in Asia:	No. 57	Medium-term Growth-Stabilization
140. 40	Issues and Areas of Reforms	2.0. 0.	Relationship in Asian Developing Countries
	Issues and Areas of Reforms I. Ali, November 1988		and Some Policy Considerations
Nt. 44	Service Trade and Asian Developing Economies		-Yun-Hwan Kim. February 1993
No. 44	Service frage and Asian Developing Decinomies	No. 58	Urbanization, Population Distribution,
	-M.G. Quibria, October 1989	140. 30	and Economic Development in Asia
No. 45	A Review of the Economic Analysis of Power		-Ernesto M. Pernia, February 1993
	Projects in Asia and Identification of Areas	No. 59	The Need for Fiscal Consolidation in Nepal:
	of Improvement	140. 55	The Results of a Simulation
	-I. Ali, November 1989		-Filippo di Mauro and Ronald Antonio Butiong,
No. 46	Growth Perspective and Challenges for Asia:		July 1993
	Areas for Policy Review and Research	No. 60	A Computable General Equilibrium Model
	—I. Ali, November 1989	140. 60	of Nepal
No. 47	An Approach to Estimating the Poverty		—Timothy Buehrer and Filippo di Mauro, October
	Alleviation Impact of an Agricultural Project		1993
	—I. Ali, January 1990	NI. 01	The Role of Government in Export Expansion
No. 48	Economic Growth Performance of Indonesia,	No. 61	
	the Philippines, and Thailand:		in the Republic of Korea: A Revisit
	The Human Resource Dimension	NT 00	—Yun-Hwan Kim, February 1994
	-E.M. Pernia, January 1990	No. 62	Rural Reforms, Structural Change,
No. 49	Foreign Exchange and Fiscal Impact of a Project:		and Agricultural Growth in
	A Methodological Framework for Estimation		the People's Republic of China
	—I. Ali, February 1990	NT 00	—Bo Lin, August 1994
No. 50	Public Investment Criteria: Financial	No. 63	Incentives and Regulation for Pollution Abatement
	and Economic Internal Rates of Return		with an Application to Waste Water Treatment
	—I. Ali, April 1990		-Sudipto Mundle, U. Shankar, and Shekhar
No. 51	Evaluation of Water Supply Projects:		Mehta, October 1995
	An Economic Framework	No. 64	Saving Transitions in Southeast Asia
	-Arlene M. Tadle, June 1990		-Frank Harrigan, February 1996
No. 52	Interrelationship Between Shadow Prices, Project	No. 65	Total Factor Productivity Growth in East Asia:
	Investment, and Policy Reforms:		A Critical Survey
	An Analytical Framework		-Jesus Felipe, September 1997
	—I. Ali, November 1990	No. 66	Foreign Direct Investment in Pakistan:
No. 53	Issues in Assessing the Impact of Project		Policy Issues and Operational Implications
	and Sector Adjustment Lending		—Ashfaque H. Khan and Yun-Hwan Kim, July
	—I. Ali, December 1990		1999
No. 54	Some Aspects of Urbanization	No. 67	Fiscal Policy, Income Distribution and Growth
	and the Environment in Southeast Asia		—Sailesh K. Jha, November 1999
	—Ernesto M. Pernia, January 1991		

ECONOMIC STAFF PAPERS (ES)

No.	1	International Reserves: Factors Determining Needs and Adequacy	No. 21	—Harry T. Oshima, October 1983 The Significance of Off-Farm Employment and Incomes in Post-War East Asian Growth
No.	2	-Evelyn Go, May 1981 Domestic Savings in Selected Developing	N- 00	-Harry T. Oshima, January 1984
		Asian Countries —Basil Moore, assisted by A.H.M. Nuruddin	No. 22	Income Distribution and Poverty in Selected Asian Countries —John Malcolm Dowling, Jr., November 1984
No.	3	Chowdhury, September 1981 Changes in Consumption, Imports and Exports	No. 23	ASEAN Economies and ASEAN Economic
		of Oil Since 1973: A Preliminary Survey of the Developing Member Countries		Cooperation —Narongchai Akrasanee, November 1984
		of the Asian Development Bank —Dal Hyun Kim and Graham Abbott, September	No. 24	Economic Analysis of Power Projects —Nitin Desai, January 1985
		1981	No. 25	Exports and Economic Growth in the Asian Region
No.	4	By-Passed Areas, Regional Inequalities, and Development Policies in Selected	No. 26	—Pradumna Rana, February 1985 Patterns of External Financing of DMCs
		Southeast Asian Countries —William James, October 1981	No. 27	—E. Go, May 1985 Industrial Technology Development
No.	5	Asian Agriculture and Economic Development	• • • • • • • • • • • • • • • • • • • •	the Republic of Korea
No.	6		No. 28	—S.Y. Lo, July 1985 Risk Analysis and Project Selection:
		An Analysis of Recent Trends —A.H.M. Nuruddin Chowdhury and J. Malcolm		A Review of Practical Issues —J.K. Johnson, August 1985
	_	Dowling, March 1982	No. 29	Rice in Indonesia: Price Policy and Comparative
No.	7	Industrial Growth and Employment in Developing Asian Countries: Issues and		Advantage —I. Ali, January 1986
		Perspectives for the Coming Decade —Ulrich Hiemenz, March 1982	No. 30	Effects of Foreign Capital Inflows on Developing Countries of Asia
No.	8	Petrodollar Recycling 1973-1980.		—Jungsoo Lee, Pradumna B. Rana, and Yoshihiro Iwasaki, April 1986
		Part 1: Regional Adjustments and the World Economy	No. 31	Economic Analysis of the Environmental
No.	9	—Burnham Campbell, April 1982 Developing Asia: The Importance		Impacts of Development Projects —John A. Dixon et al., EAPI, East-West Center,
		of Domestic Policies	No. 32	August 1986 Science and Technology for Development:
		—Economics Office Staff under the direction of Seiji Naya, May 1982	140. 52	Role of the Bank
No.	10	Financial Development and Household Savings: Issues in Domestic Resource	No. 33	—Kedar N. Kohli and Ifzal Ali, November 1986 Satellite Remote Sensing in the Asian
		Mobilization in Asian Developing Countries —Wan-Soon Kim, July 1982		and Pacific Region —Mohan Sundara Rajan, December 1986
No.	11	Industrial Development: Role of Specialized	No. 34	Changes in the Export Patterns of Asian and
		Financial Institutions —Kedar N. Kohli, August 1982		Pacific Developing Countries: An Empirical Overview
No.	12	Petrodollar Recycling 1973-1980. Part II: Debt Problems and an Evaluation	No. 35	—Pradumna B. Rana, January 1987 Agricultural Price Policy in Nepal
		of Suggested Remedies		-Gerald C. Nelson, March 1987
No.	13	-Burnham Campbell, September 1982 Credit Rationing, Rural Savings, and Financial	No. 36	Implications of Falling Primary Commodity Prices for Agricultural Strategy in the Philippines
		Policy in Developing Countries —William James, September 1982	No. 37	—Ifzal Ali, September 1987 Determining Irrigation Charges: A Framework
No.	14	Small and Medium-Scale Manufacturing		—Prabhakar B. Ghate, October 1987 The Role of Fertilizer Subsidies in Agricultural
		Establishments in ASEAN Countries: Perspectives and Policy Issues	No. 38	Production: A Review of Select Issues
No.	15	—Mathias Bruch and Ulrich Hiemenz, March 1983 Income Distribution and Economic	No. 39	—M.G. Quibria, October 1987 Domestic Adjustment to External Shocks
• • • •		Growth in Developing Asian Countries		in Developing Asia —Jungsoo Lee, October 1987
No.	16	—J. Malcolm Dowling and David Soo, March 1983 Long-Run Debt-Servicing Capacity of	No. 40	Improving Domestic Resource Mobilization
		Asian Developing Countries: An Application of Critical Interest Rate Approach		through Financial Development: Indonesia —Philip Erquiaga, November 1987
		-Jungsoo Lee, June 1983	No. 41	Recent Trends and Issues on Foreign Direct Investment in Asian and Pacific Developing
No.	. 17	External Shocks, Energy Policy, and Macroeconomic Performance of Asian		Countries
		Developing Countries: A Policy Analysis —William James, July 1983	No. 42	—P.B. Rana, March 1988 Manufactured Exports from the Philippines:
	. 18	The Impact of the Current Exchange Rate		A Sector Profile and an Agenda for Reform —I. Ali, September 1988
		System on Trade and Inflation of Selected Developing Member Countries	No. 43	A Framework for Evaluating the Economic
	. 19	Pradumna Rana, September 1983 Asian Agriculture in Transition: Key Policy Issues		Benefits of Power Projects —I. Ali, August 1989
		-William James, September 1983	No. 44	Promotion of Manufactured Exports in Pakistan —Jungsoo Lee and Yoshihiro Iwasaki, September
NO	. 20	The Transition to an Industrial Economy in Monsoon Asia		1989

- No. 45 Education and Labor Markets in Indonesia: A Sector Survey
 —Ernesto M. Pernia and David N. Wilson, September 1989
- No. 46 Industrial Technology Capabilities and Policies in Selected ADCs —Hiroshi Kakazu, June 1990
- No. 47 Designing Strategies and Policies for Managing Structural Change in Asia —Ifzal Ali. June 1990
- No. 48 The Completion of the Single European Community
 Market in 1992: A Tentative Assessment of its
 Impact on Asian Developing Countries
 —J.P. Verbiest and Min Tang, June 1991
- No. 49 Economic Analysis of Investment in Power Systems

 —Ifzal Ali, June 1991
- No. 50 External Finance and the Role of Multilateral Financial Institutions in South Asia:
 Changing Patterns, Prospects, and Challenges
 —Jungsoo Lee, November 1991
- No. 51 The Gender and Poverty Nexus: Issues and Policies
 —M.G. Quibria, November 1993
- No. 52 The Role of the State in Economic Development:
 Theory, the East Asian Experience,
 and the Malaysian Case
 —Jason Brown, December 1993

- No. 53 The Economic Benefits of Potable Water Supply Projects to Households in Developing Countries

 —Dale Whittington and Venkateswarlu Swarna, January 1994
- No. 54 Growth Triangles: Conceptual Issues and Operational Problems

 —Min Tang and Myo Thant, February 1994
- No. 55 The Emerging Global Trading Environment and Developing Asia

 —Arvind Panagariya, M.G. Quibria, and Narhari Rao. July 1996
- No. 56 Aspects of Urban Water and Sanitation in the Context of Rapid Urbanization in Developing Asia

 —Ernesto M. Pernia and Stella LF. Alabastro, September 1997
- No. 57 Challenges for Asia's Trade and Environment

 —Douglas H. Brooks, January 1998
- No. 58 Economic Analysis of Health Sector Projects-A Review of Issues, Methods, and Approaches
 —Ramesh Adhikari, Paul Gertler, and Anneli
 Lagman, March 1999
- No. 59 The Asian Crisis: An Alternate View
 —Rajiv Kumar and Bibek Debroy, July 1999
- No. 60 Social Consequences of the Financial Crisis in
 Asia
 —James C. Knowles, Ernesto M. Pernia, and Mary
 Racelis, November 1999

OCCASIONAL PAPERS (OP)

- No. 1 Poverty in the People's Republic of China:
 Recent Developments and Scope
 for Bank Assistance
 —K.H. Moinuddin, November 1992
- No. 2 The Eastern Islands of Indonesia: An Overview of Development Needs and Potential

 —Brien K. Parkinson, January 1993
- No. 3 Rural Institutional Finance in Bangladesh and Nepal: Review and Agenda for Reforms

 —A.H.M.N. Chowdhury and Marcelia C. Garcia, November 1993
- No. 4 Fiscal Deficits and Current Account Imbalances of the South Pacific Countries:

 A Case Study of Vanuatu

 —T.K. Jayaraman, December 1993
- No. 5 Reforms in the Transitional Economies of Asia

 —Pradumna B. Rana, December 1993
- No. 6 Environmental Challenges in the People's Republic of China and Scope for Bank Assistance
 —Elisabetta Capannelli and Omkar L. Shrestha,
 December 1993
- No. 7 Sustainable Development Environment and Poverty Nexus

 -K.F. Jalal, December 1993
- No. 8 Intermediate Services and Economic
 Development: The Malaysian Example
 —Sutanu Behuria and Rahul Khullar, May 1994
- No. 9 Interest Rate Deregulation: A Brief Survey of the Policy Issues and the Asian Experience

 —Carlos J. Glower, July 1994
- No. 10 Some Aspects of Land Administration in Indonesia: Implications for Bank Operations
 —Sutanu Behuria, July 1994
- No. 11 Demographic and Socioeconomic Determinants of Contraceptive Use among Urban Women in the Melanesian Countries in the South Pacific:

 A Case Study of Port Vila Town in Vanuatu

 —T.K. Jayaraman, February 1995

- No. 12 Managing Development through Institution Building — Hilton L. Root, October 1995
- No. 13 Growth, Structural Change, and Optimal Poverty Interventions
 —Shiladitya Chatterjee, November 1995
- No. 14 Private Investment and Macroeconomic Environment in the South Pacific Island Countries: A Cross-Country Analysis

 —T.K. Jayaraman, October 1996
- No. 15 The Rural-Urban Transition in Viet Nam:
 Some Selected Issues
 —Sudipto Mundle and Brian Van Arkadie, October
 1997
- No. 16 A New Approach to Setting the Future
 Transport Agenda
 —Roger Allport, Geoff Key, and Charles Melhuish,
 June 1998
- No. 17 Adjustment and Distribution:
 The Indian Experience
 —Sudipto Mundle and V.B. Tulasidhar, June 1998
- No. 18 Tax Reforms in Viet Nam: A Selective Analysis
 —Sudipto Mundle, December 1998
- No. 19 Surges and Volatility of Private Capital Flows to Asian Developing Countries: Implications for Multilateral Development Banks —Pradumna B. Rana, December 1998
- No. 20 The Millennium Round and the Asian Economies:
 An Introduction
 —Dilip K. Das, October 1999
- No. 21 Occupational Segregation and the Gender
 Earnings Gap

 —Joseph E. Zveglich, Jr. and Yana van der Meulen
- Rodgers, December 1999

 No. 22 Information Technology: Next Locomotive of Growth?

 —Dilip K. Das, June 2000

STATISTICAL REPORT SERIES (SR)

No. 1	Estimates of the Total External Debt of the Developing Member Countries of ADB:		in Asian Developing Countries, 1985 —Jungsoo Lee and I.P. David, April 1987
	1981-1983	No. 10	A Survey of the External Debt Situation
	—I.P. David, September 1984	No. 10	in Asian Developing Countries, 1986
No. 2	Multivariate Statistical and Graphical		—Jungsoo Lee and I.P. David, April 1988
NO. 2	Classification Techniques Applied	Nt. 11	Changing Pattern of Financial Flows to Asian
	to the Problem of Grouping Countries	No. 11	and Pacific Developing Countries
	—I.P. David and D.S. Maligalig, March 1985		
31 0	Gross National Product (GNP) Measurement		Jungsoo Lee and I.P. David, March 1989
No. 3	Issues in South Pacific Developing Member	No. 12	The State of Agricultural Statistics in
	Countries of ADB		Southeast Asia
	*******		—I.P. David, March 1989
	—S.G. Tiwari, September 1985	No. 13	A Survey of the External Debt Situation
No. 4	Estimates of Comparable Savings in Selected DMCs		in Asian and Pacific Developing Countries:
	-Hananto Sigit, December 1985		Jungsoo Lee and I.P. David, July 1989
No. 5	Keeping Sample Survey Design	No. 14	A Survey of the External Debt Situation in
110. 0	and Analysis Simple	140. 14	Asian and Pacific Developing Countries: 1988-1989
	—I.P. David, December 1985		Jungsoo Lee, May 1990
No. 6	External Debt Situation in Asian	No. 15	A Survey of the External Debt Situation
110. 0	Developing Countries	140. 10	in Asian and Pacific Developing Countries: 1989-
	—I.P. David and Jungsoo Lee, March 1986		1992
No. 7	Study of GNP Measurement Issues in the		-Min Tang, June 1991
110. 1	South Pacific Developing Member Countries.	No. 16	Recent Trends and Prospects of External Debt
	Part I: Existing National Accounts	140. 10	Situation and Financial Flows to Asian
	of SPDMCs-Analysis of Methodology		and Pacific Developing Countries
	and Application of SNA Concepts		-Min Tang and Aludia Pardo, June 1992
	-P. Hodgkinson, October 1986	No. 17	Purchasing Power Parity in Asian Developing
No. 8	Study of GNP Measurement Issues in the South	140. 17	Countries: A Co-Integration Test
140. 0	Pacific Developing Member Countries.		—Min Tang and Ronald Q. Butiong, April 1994
	Part II: Factors Affecting Intercountry	No. 18	Capital Flows to Asian and Pacific Developing
	Comparability of Per Capita GNP	140. 10	Countries: Recent Trends and Future Prospects
	—P. Hodgkinson, October 1986		-Min Tang and James Villafuerte, October 1995
No. 9	Survey of the External Debt Situation		-izas lung una vames tampacine, conter love

SERIALS

(Available commercially through ADB Office of External Relations)

- Asian Development Outlook (ADO; annual) \$36.00 (paperback)
 Key Indicators of Developing Asian and Pacific Countries (KI; annual) \$35.00 (paperback)
- 3. Asian Development Review (ADR; semiannual) \$5.00 per issue; \$8.00 per year (2 issues)

SPECIAL STUDIES, CO-PUBLISHED

(Available commercially through Oxford University Press Offices, Edward Elgar Publishing, and Palgrave MacMillan)

FROM OXFORD UNIVERSITY PRESS:
Oxford University Press (China) Ltd
18th Floor, Warwick House East
Taikoo Place, 979 King's Road
Quarry Bay, Hong Kong
Tel (852) 2516 3222
Fax (852) 2565 8491
E-mail: webmaster@oupchina.com.hk
Web: www.oupchina.com.hk

- Informal Finance: Some Findings from Asia Prabhu Ghate et. al., 1992 \$15.00 (paperback)
- Mongolia: A Centrally Planned Economy in Transition
 Asian Development Bank, 1992
 \$15.00 (paperback)
- 3. Rural Poverty in Asia, Priority Issues and Policy Options

Edited by M.G. Quibria, 1994 \$25.00 (paperback)

- 4. Growth Triangles in Asia: A New Approach to Regional Economic Cooperation

 Edited by Myo Thant, Min Tang, and Hiroshi Kakazu

 1st ed., 1994 \$36.00 (hardbound)

 Revised ed., 1998 \$55.00 (hardbound)
- Urban Poverty in Asia: A Survey of Critical Issues Edited by Ernesto Pernia, 1994 \$18.00 (paperback)
- Critical Issues in Asian Development: Theories, Experiences, and Policies Edited by M.G. Quibria, 1995 \$15.00 (paperback)
 \$36.00 (hardbound)
- 7. Financial Sector Development in Asia
 Edited by Shahid N. Zahid, 1995
 \$50.00 (hardbound)
- Financial Sector Development in Asia: Country Studies
 Edited by Shahid N. Zahid, 1995
 \$55.00 (hardbound)
- Fiscal Management and Economic Reform in the People's Republic of China Christine P.W. Wong, Christopher Heady, and Wing T. Woo, 1995 \$15.00 (paperback)
- 10. From Centrally Planned to Market Economies:
 The Asian Approach

Edited by Pradumna B. Rana and Naved Hamid, 1995 Vol. 1: Overview

\$36.00 (hardbound)

Vol. 2: People's Republic of China and Mongolia \$50.00 (hardbound)

Vol. 3: Lao PDR, Myanmar, and Viet Nam \$50.00 (hardbound)

- Current Issues in Economic Development:
 An Asian Perspective
 Edited by M.G. Quibria and J. Malcolm Dowling, 1996
 \$50.00 (hardbound)
- 12. The Bangladesh Economy in Transition Edited by M.G. Quibria, 1997 \$20.00 (hardbound)
- The Global Trading System and Developing Asia Edited by Arvind Panagariya, M.G. Quibria, and Narhari Rao, 1997 \$55.00 (hardbound)
- Social Sector Issues in Transitional Economies of Asia
 Edited by Douglas H. Brooks and Myo Thant, 1998
 \$25.00 (paperback)
 \$55.00 (hardbound)
- Intergovernmental Fiscal Transfers in Asia: Current Practice and Challenges for the Future Edited by Yun-Hwan Kim and Paul Smoke, 2003 \$15.00 (paperback)
- Local Government Finance and Bond Markets Edited by Yun-Hwan Kim, 2003 \$15.00 (paperback)

FROM EDWARD ELGAR:
Marston Book Services Limited
PO Box 269, Abingdon
Oxon OX14 4YN, United Kingdom
Tel +44 1235 465500
Fax +44 1235 465555
Email: direct.order@marston.co.uk
Web: www.marston.co.uk

 Reducing Poverty in Asia: Emerging Issues in Growth, Targeting, and Measurement Edited by Christopher M. Edmonds, 2003

FROM PALGRAVE MACMILLAN:
Palgrave Macmillan Ltd
Houndmills, Basingstoke
Hampshire RG21 6XS, United Kingdom
Tel: +44 (0)1256 329242
Fax: +44 (0)1256 479476
Email: orders@palgrave.com
Web: www.palgrave.com/home/

- Labor Markets in Asia: Issues and Perspectives Edited by Jesus Felipe and Rana Hasan, 2006
- Competition Policy and Development in Asia Edited by Douglas H. Brooks and Simon Evenett, 2005
- Managing FDI in a Globalizing Economy
 Asian Experiences
 Edited by Douglas H. Brooks and Hal Hill, 2004
- Poverty, Growth, and Institutions in Developing Asia Edited by Ernesto M. Pernia and Anil B. Deolalikar, 2003

SPECIAL STUDIES. IN-HOUSE

(Available commercially through ADB Office of External Relations)

1. Rural Poverty in Developing Asia Edited by M.G. Quibria

Vol. 1: Bangladesh, India, and Sri Lanka, 1994 \$35.00 (paperback)

Vol. 2: Indonesia, Republic of Korea, Philippines, and Thailand, 1996

\$35.00 (paperback)

2. Gender Indicators of Developing Asian and Pacific Countries

Asian Development Bank, 1993 \$25.00 (paperback)

3. External Shocks and Policy Adjustments: Lessons from the Gulf Crisis

Edited by Naved Hamid and Shahid N. Zahid, 1995 \$15.00 (paperback)

4. Indonesia-Malaysia-Thailand Growth Triangle:
Theory to Practice

Edited by Myo Thant and Min Tang, 1996 \$15.00 (paperback)

 Emerging Asia: Changes and Challenges Asian Development Bank, 1997
 \$30.00 (paperback)

\$30.00 (paperback)
6. Asian Exports
Edited by Dilip Das, 1999
\$35.00 (paperback)
\$55.00 (hardbound)

Development of Environment Statistics in Developing
 Asian and Pacific Countries
 Asian Development Bank, 1999
 \$30.00 (paperback)

8. Mortgage-Backed Securities Markets in Asia Edited by S.Ghon Rhee & Yutaka Shimomoto, 1999 \$35.00 (paperback)

9. Rising to the Challenge in Asia: A Study of Financial Markets

Asian Development Bank

Vol. 1: An Overview, 2000 \$20.00 (paperback)

Vol. 2: Special Issues, 1999 \$15.00 (paperback)

Vol. 3: Sound Practices, 2000 \$25.00 (paperback)

Vol. 4: People's Republic of China, 1999 \$20.00 (paperback)

Vol. 5: India, 1999 \$30.00 (paperback)

Vol. 6: Indonesia, 1999 \$30.00 (paperback)

Vol. 7: Republic of Korea, 1999 \$30.00 (paperback)

Vol. 8: Malaysia, 1999 \$20.00 (paperback)

Vol. 9: Pakistan, 1999 \$30.00 (paperback)

Vol. 10: Philippines, 1999 \$30.00 (paperback)

Vol. 11: Thailand, 1999 \$30.00 (paperback)

Vol. 12: Socialist Republic of Viet Nam, 1999 \$30.00 (paperback)

Corporate Governance and Finance in East Asia:
 A Study of Indonesia, Republic of Korea, Malaysia, Philippines and Thailand

J. Zhuang, David Edwards, D. Webb, & Ma. Virginita Capulong Vol. 1: A Consolidated Report, 2000 \$10.00 (paperback) Vol. 2: Country Studies, 2001 \$15.00 (paperback)

Financial Management and Governance Issues
 Asian Development Bank, 2000
 Cambodia \$10.00 (paperback)
 People's Republic of China \$10.00 (paperback)
 Mongolia \$10.00 (paperback)
 Pakistan \$10.00 (paperback)
 Papua New Guinea \$10.00 (paperback)
 Uzbekistan \$10.00 (paperback)

Viet Nam \$10.00 (paperback)
Selected Developing Member Countries \$10.00 (paperback)

 Government Bond Market Development in Asia Edited by Yun-Hwan Kim, 2001 \$25.00 (paperback)

 Intergovernmental Fiscal Transfers in Asia: Current Practice and Challenges for the Future Edited by Paul Smoke and Yun-Hwan Kim, 2002 \$15.00 (paperback)

 Guidelines for the Economic Analysis of Projects Asian Development Bank, 1997 \$10.00 (paperback)

 Guidelines for the Economic Analysis of Telecommunications Projects Asian Development Bank, 1997 \$10.00 (paperback)

 Handbook for the Economic Analysis of Water Supply Projects Asian Development Bank, 1999 \$10.00 (hardbound)

Handbook for the Economic Analysis of Health Sector Projects
 Asian Development Bank, 2000
 \$10.00 (paperback)

 Handbook for Integrating Povery Impact Assessment in the Economic Analysis of Projects Asian Development Bank, 2001 \$10.00 (paperback)

 Handbook for Integrating Risk Analysis in the Economic Analysis of Projects Asian Development Bank, 2002

\$10.00 (paperback)
20. Handbook on Environment Statistics
Asian Development Bank, 2002
\$10.00 (hardback)

21. Defining an Agenda for Poverty Reduction, Volume 1

Edited by Christopher Edmonds and Sara Medina,
2002

\$15.00 (paperback)

Defining an Agenda for Poverty Reduction, Volume 2
 Edited by Isabel Ortiz, 2002
 \$15.00 (paperback)

23. Economic Analysis of Policy-based Operations: Key Dimensions
Asian Development Bank, 2003
\$10.00 (paperback)